The relationship among generational cohorts, tenure, job categories, and employee readiness for organizational change in a healthcare environment: A Quantitative Study

by

Jerry S.K. Adatsi Newnan, Georgia

A dissertation submitted in partial fulfillment of the requirements for the degree of

# DOCTOR OF BUSINESS ADMINISTRATION

Graduate Faculty of the College of Business George Fox University

**Dissertation Committee:** 

Dr. Paul Shelton, Chair Dr. Dirk Barram, Member Dr. Brad Jensen, Member

> Newberg, Oregon January 2020



# **Dissertation Completion Approval Doctor of Business Administration**

Student Name: Jerry Adatsi

Student ID#: 1783801

Cohort #: \_\_\_\_\_

Concentration: MGMT

Project Title:

The Relationship Among Generational Cohorts, Tenure, Job Categories and Employee Readiness for Organizational Change in a Healthcare Environment: A Quantitative Study

im

has been approved for the Doctor of Business Administration Program at George Fox University as a dissertation for the DBA degree.

Approval Signatures:

Chair

Dr. Paul Shelton, PhD Chair (print)

Member

1-24-20 Date

Date 1/24/00

Date

Dr. Dirk Barram, PhD Member (print)

Member

Dr. Brad Jensen, PhD Member (print)

\_\_\_\_ Date 1/24/20

Dr. Justine Haigh, PhD Director, DBA Program

Jerry S.K. Adatsi, MBA

Jadatsi15@georgefox.edu

© 2020 Jerry S.K. Adatsi All Rights Reserved

# Dedication

I dedicate this dissertation to my Lord God, Jesus Christ, who has provided me with faith, strength, and a loving family, without whom I would not have been healed from cancer or achieved anything in life, let alone this doctoral degree. With gratitude, I dedicate this effort of my learning to the memory of my late grandmother, Nana Grace Yawa Darkey, The Queen Mother of Peki Avertile, Ghana. (Màmá, I kept my promise and achieved this doctoral degree. I will never stop learning. Thank you, Màmá.) And to my wife Tiffani (your love and devotion encouraged me to achieve my dreams); my two extraordinary sons, Jeremiah and Jonathan with the hope that they understand that learning never ends even with life's many obstacles; and that faith in the Lord God, Jesus Christ and obedience to Him is life's greatest achievement. This study is dedicated to those teachers who inspired me to love learning, specifically, my high school English teacher and "adoptive" mother, Susan Rawlston: words cannot express how grateful I am to you for seeing the best in me, supporting me through all my educational endeavors, and standing by me as I fought and won my battle with cancer. Finally, just as important, this study is dedicated to those who were not teachers but taught me something about life or something about myself.

#### Acknowledgments

All praise, glory, honor, and thanksgiving to the Almighty God, the Lord Jesus Christ, for giving me life and the resolve to pursue a doctoral study. While completing a dissertation is considered an individual accomplishment, I believe, regardless of prevailing sentiments, that solitary endeavors and achievements do not exist as there is no such thing as "self-made." Many hands guided and supported me along in this effort. I want to acknowledge my wife, Tiffani Adatsi, NP; my two sons Jeremiah and Jonathan Adatsi; my church, a house of the Living God, Church of Jesus Christ; and all my friends for their prayers, unwavering support, and kind words of encouragement throughout this endeavor.

It is with a grateful and humble heart that I acknowledge the chair of my dissertation committee, Dr. Paul Shelton, who mentored and guided me through this dissertation process. I was blessed to interview with Dr. Shelton when I applied for admission into the DBA program at George Fox University and recall our conversation surrounding the topic of scholar-practitioner, and knew then that I wanted him to chair my committee. Dr. Shelton, thank you for setting me on the right course from the beginning. I am incredibly grateful to the members of my committee, Dr. Dirk Barram and Dr. Brad Jensen. I appreciated the encouragement, support, and thoroughly insightful feedback, which allowed me to view my dissertation from another perspective. Your help made this journey more manageable and my degree attainable.

Finally, I would like to acknowledge my cohort (George Fox University Doctor of Business Administration Cohort 11). I am grateful to be included in this group of scholar-

v

practitioners. Thank you for the insightful class discussions, presentations, support, and encouragement. To the Lord God Jesus be the Glory.

| Table of | Contents |
|----------|----------|
|----------|----------|

| Dedication                                     | iv |
|--|----|
| Acknowledgments                                | v  |
| List of Tables                                 | x  |
| List of Figures                                | xi |
| ABSTRACT                                       | 1  |
| CHAPTER 1- INTRODUCTION                        | 2  |
| Background of the study                        | 4  |
| Statement of the research problem              | 6  |
| Purpose of the research                        | 8  |
| Research question and hypotheses               | 11 |
| The Omnibus research question and hypothesis   | 11 |
| Definition of terms                            | 14 |
| Operational definitions                        | 16 |
| Acronyms                                       | 18 |
| Assumptions, limitations, and delimitations    | 18 |
| The significance of the study                  | 20 |
| The theoretical and conceptual framework       | 21 |
| Researcher's perspective                       | 22 |
| Summary  | 23 |
| CHAPTER 2- LITERATURE REVIEW                   |    |
| Historical view of organizational change       | 25 |
| The cost and failures of organizational change | 28 |
| Readiness for organizational change            |    |
| Resistance to organizational change            |    |
| Employee response to organizational change     | 34 |
| Positive view of change resistance             | 35 |
|  |    |

| Summary   |    |
|---|----|
| Generational Cohorts Theory                       |    |
| Historical view of generation theory              |    |
| Description of generations                        |    |
| Generational differences and similarities         | 45 |
| Generational response to change                   | 45 |
| The impact of tenure and job categories           | 46 |
| Overview of correlational analysis model          | 47 |
| Summary   | 48 |
| CHAPTER 3- METHOD                                 |    |
|   |    |
| Research rationale                                | 52 |
| Research design                                   | 53 |
| Research questions, sub-questions, and hypothesis | 53 |
| Population and sample                             | 57 |
| Informed consent                                  | 59 |
| Confidentiality                                   | 60 |
| Measures  | 60 |
| Procedure   | 62 |
| Data analysis                                     | 63 |
| Ethical considerations                            | 64 |
| Summary   | 65 |
| CHAPTER 4- RESULTS                                |    |
|   |    |
| Data collection                                   |    |
| Description of the population and sample          |    |
| Sample frame and sample size                      | 68 |
| Effect size                                       |    |
| Description of participants                       | 70 |
| Research design and methodology                   | 73 |
| Measures  | 74 |
| Instrument reliability and viability              | 75 |
| Details of data analysis and results              | 77 |
| Means and standard deviations                     | 78 |

| Summary  |
|--|
| CHAPTER 5- DISCUSSION                                    |
|  |
| Significance of the study                                |
| Discussion of findings                                   |
| Hypothesis one $(H_0)$ and Hypothesis two $(H_01)$ 9     |
| Negative relationships: TE, GENCO, and ROC9              |
| Hypothesis three (H <sub>0</sub> 2)9                     |
| Summary  |
| Literature support                                       |
| Implications of the study9                               |
| Limitations of the study9                                |
| Recommendations  |
| Conclusions  |
| REFERENCES 100   |
| APPENDICES   |
| Appendix A: Informed consent11                           |
| Appendix B: Readiness for organizational change survey12 |
| Appendix C: Permission for the use of copyright content  |

# List of Tables

| Table 1: Population by Gender and Generational Cohorts 71                               |
|---|
| Table 2: Population by Tenure level and Position Categories 71                          |
| Table 3: Reliability Analysis of OCQ-R Scale Instrument                                 |
| Table 4: Mean and Standard Deviation of Variables and the Dimensions of Change          |
| Readiness   |
| Table 5: Correlational Analysis of Tenure Level, Position Category, Generations Cohort, |
| and Change Readiness  |
| Table 6: Generational Cohorts Correlated with Employee Change Readiness    80           |
| Table 7: Generation Cohorts Correlated with Position Category 81                        |
| Table 8: Correlation Analysis of Generation Cohorts and Tenure Level 81                 |
| Table 9: Correlation Analysis of Generational Cohorts, Internal Readiness for Change,   |
| Cognitive Readiness for Change, and Emotional Readiness for Change                      |
| Table 10:Regression Analysis of Change Readiness and Generational Cohorts               |
| Table 11: General Linear Model Analysis of GENCO, TE, POC, and ROC 83                   |
| Table 12: Analysis of Variance Between the Dimensions of Readiness and Generational     |
| Cohorts   |

# List of Figures

| Figure 1: Conceptual Framework: Generational Cohort on Employee Readiness for |   |
|---|---|
| Organizational Change moderated by Tenure Level and Position Category         | 2 |

# ABSTRACT

Internal and external factors constantly influence and pressure healthcare organizations to adapt, change, or respond to changes in the modern global business environment. When facing the challenge of change initiatives, healthcare organizations depend on the flexibility of employees to successfully navigate the change process as there is the expectation that their employees will be ready for a change. Extant literature, however, suggests that more than 70% of change initiatives undertaken by businesses fail, and the high cost of failure is due to the resistance of employees to change and their lack of change readiness. With continued interest by healthcare organizations to understand how to navigate the organizational change process and the growing generational diversity in the modern work environment, this study evaluated the relationship of generational diversity on employees' readiness for organizational change; the purpose of this study is to determine if there is a relationship among generational cohorts and change readiness. This research also examined the relationship of employee tenure and position and their interactive effects with generational cohorts to moderate readiness for change. Results indicate that a positive relationship exists between generational cohort characteristics, tenure, position category, and the dimensions of change readiness. The positive relationships, however, were not statistically significant between variables.

# **CHAPTER 1- INTRODUCTION**

Technological advancements, a changing workforce, the lack of change readiness, competitive pressures, and globalization are among factors which contribute to the need for change to be planned, initiated, and implemented within organizations, a process that requires engagement from all organizational members and allows companies to compete (Abdel-Ghany, 2014; Stouten, Rousseau, & Cremer, 2018). Organizations often struggle to create successful and sustainable change although change inevitably impacts organizational viability (Armenakis, Harris, & Mossholder, 1993; Kotter, 1996; Stouten, Rousseau, & Cremer, 2018). Organizational change (OC) "represents the conversion of the dominant culture of the organization to a more flexible, complex, and multi-level culture" (Caliskan & Isik, 2016, p. 405). Change is exercised when an organization deliberately alters its activities from a present state to a future state, defined by the difference in the form, quality, or state of an organization; the planned and unplanned process of altering the way things are done in organizations can be incremental, radical, or transformative (Beer & Walton, 1987; Poole, 1998; Stouten, et al., 2018; Van de Ven & Poole, 1995).

Benzer, Charns, Hamdan, and Afable (2017) argue that the concept of organizational change is difficult to define and is, therefore, contextualized on an organizational level. However, organizational change is explained by the change in individuals within the organization (Abdel-Ghany, 2014) and influenced by individuallevel readiness for change (Benzer, et al., 2017). Armenakis et al. (1993) and Bouckenooghe, Devos, and Broeck (2009) found that the response to change informs an employees' readiness for organizational change. The readiness for organizational change

predicts behaviors such as ambivalence, support, or resistance (Armenakis et al., 1993; Oreg & Sverdlik, 2011), and negative responses such as resistance to change have been attributed to the significantly high rate of organizational change failures. Burnes (2004) concludes, as a result of a literature review, that change failure rate, as a result of organizational change resistance, is higher than 70%. Decker, Durand, Mayfield, McCormack, Skinner, and Perdue (2012) confirmed this argument with the assertion that the rate of change failure is much closer to 93%. Since organizational change provokes adverse reactions among members of an organization and indicates a lack of readiness for change, the changing generational diversity of the modern workforce presents an additional dimension to change resistance and change failure (Higgs & Rowland, 2005; Kotter, 1996; Warrick, 2009). Research (Armenakis et al., 1993; Benzer, et al., 2017; Bouckenooghe et al., 2009; Stouten, et al., 2018; Oreg, Vakola, & Armenakis, 2011) confirms that employee readiness for change remains a persistent problem for organizations to address.

Researchers in the field of management have been increasingly examining employees' readiness for change and how their responses impact organizational change initiatives (Bouckenooghe et al., 2009; Kotter & Cohen, 2002; Shin, Taylor, & Seo, 2012; Stouten, et al., 2018). Additionally, the attitudes of the employee towards change implementation and the readiness of change in a multifaceted acute care hospital have been studied (Sharma, Hernschmidt, Claes, Batchnick, De Geest, & Simon, 2018). In this chapter, the background and purpose of the study are presented, as well as the related research questions and the hypotheses. A definition of frequently used terms and an explanation of acronyms that are used in this study are provided, and the assumptions and

limitations of the study are discussed. The significance of the study and the researcher's perspective are also presented.

# **Background of the study**

Change readiness is a critical component of organizational development and growth. The advent of external forces makes it difficult for organizations to manage change initiatives to ensure success. Some studies in existence suggests that more than 70% of change initiatives fail, resulting in the loss of billions of dollars in operation capital annually; failure is attributed to the lack of employee readiness and resistance to the initiated changes (Bateh, Casteneda, & Farah, 2013; Burke, 2013; Burnes 2004; Higgs & Rowland, 2005; Kotter, 1996; Miller, 2002; Pellettiere, 2006; Strebel, 2009; Warrick, 2009). Besides the negative responses exhibited by employees towards change, external factors such as technological advances, political mandates and policies, cultural shifts, and emerging generational diversities placed an undue pressure on companies in a state of continuous change (Axley & McMahon, 2006; Becker, Lazaric, Nelson, & Winter, 2005; Choi & Ruona, 2011; Vakola & Nikolaou, 2005). Further, some studies assert that employee resistance to change is impacted by responses predicated upon behaviors such as indifference towards the proposed change or outright resistance to the change. Another characteristic, such as the employee's support of change initiatives, is a predictive behavior that helps to inform the readiness for change (Armenakis, Harris, & Mossholder, 1993; Lamm & Gordon, 2010; Oreg & Sverdlik, 2011; Van Egeren, 2009).

The high cost of change initiative failures and the resistance to change is an incessant organizational issue to be resolved, demonstrating the lack of readiness for organizational change (Oreg, Vakola, and Armenakis, 2011). Organizational change

initiative failures have been attributed to the employees' resistance to change and have been linked to unreadiness for change (Armenakis et al., 1993; Bouckenooghe et al., 2009; Lamm & Gordon, 2010; Oreg & Sverdlik, 2011); as a result, management researchers have begun to study individual-level readiness for organizational change and change implementation across various industries (Choi, 2011; Sharma, et al., 2018). Bouckenooghe et al. (2009) declare that the change readiness phenomena have been investigated traditionally from a management perspective by organizational leaders and management researchers. Further, organizational change has been studied from a management perspective, and failures of change initiatives have been attributed to the organizational leadership's ability to manage the change process (Bommer, Rich, & Rubin, 2005; Caliskan & Isik, 2016; Nohe, Menges, Zhang, & Sonnatag, 2013). Recently, new areas of research have been examining the role that generational cohorts might play in employees' readiness for organizational change initiatives in healthcare environments (Ludviga & Sennikova, 2016; Sharma, et al., 2018).

Some literature has shown that five personality traits—personality, agreeableness, conscientiousness, emotional stability, and intellect— inform human behavior and provide a framework for organizing human characteristics (Armenakis et al., 1993; Goldberg, 1993). These personality traits have been linked to human behavior or characteristics (Armenakis et al., 1993), and employee behavior has been linked to organizational change success. As such, there has been considerable interest in studying the relationship between personality traits and employee behavior such as resistance to change and readiness for change. However, there is little emphasis on examining the behavioral response of generational cohort characteristics as they relate to organizational

change or readiness for organizational change in the extant literature. For example, a study by Bourne (2015) is among the limited literature that investigated the relationship between generational cohorts and their response to organizational change. Additionally, the study by Sharma, et al. (2018) provides limited research to investigate the organizational change readiness in acute care healthcare systems. The limited research of Bourne (2015) and Sharma, et al. (2018) revealed gaps in the readiness for organizational change literature. With the changing workforce in today's work environments and the persistent lack of change readiness (Bouckenooghe et al., 2009; Bourne, 2015), these gaps present the opportunity to extend the existing literature on readiness for organizational change and the generational cohort behavioral response to organizational change by examining the phenomenon among a sample of healthcare employees with different positions and tenure levels.

#### Statement of the research problem

Organizational change initiatives fail due to employee resistance and the lack of readiness for change (Bouckenooghe et al., 2009; Burnes, 2004). These studies have examined the relationship between the readiness for change of an organization's workforce and the success of change initiatives. Some existing literature on the topic of organizational change suggests that resistance to change is a behavioral response ambivalently manifested in support of or outright resistance to the change initiatives (Armenakis et al., 1993; Lamm & Gordon, 2010; Oreg & Sverdlik, 2011; Van Egeren, 2009). The principles of readiness for change theory indicates that when employees are ready for change, they will be better prepared for it and more likely to support the

initiated change (Armenakis et al., 1993). However, how different generations react and respond to change is not yet fully known.

Armenakis & Harris (2002) argue that the organizational readiness for change is highly dependent on the change readiness of the individual members of an organization's workforce. Further, Weiner (2009) asserts that experience is a contextual factor which might influence the valence for change which can be driven by position, tenure, or peer influence. Cohen (1991, 1992) asserts that there is a moderating effect of employee tenure and position groups on organizational commitment and outcomes. As such, this study seeks to determine the moderating effects of tenure and position groups on the generational cohort's readiness for change. Moreover, there is a persistent change in the workforce demographics of many organizations, a continued initiation of change processes to ensure organizational success, and a consistently high rate of change failures—a ubiquitous problem which presents the lack of understanding about how increased generational cohort diversity uniquely impacts organizational change initiatives across all industries and locations (Burnes, 2004; Decker, et al., 2012; Lesser and Rivera, 2006).

Although several studies focused on personality traits and readiness for change and the cultural character of generations, little is known about the relationship among generational cohorts and their readiness for organizational change. Since readiness for change is multi-faceted and categorized into the three dimensions of cognition, emotion, and intention (Bouckenooghe et al., 2009), the lack of literature on the generational cohorts' readiness for change and burgeoning generational diversity of the modern workforce presents an opportunity to extend discussion on the topic. The lack of a

successful organizational change model to significantly impact the service outcomes of the healthcare industry continues to emerge, although the industry undertakes many organizational change initiatives (Bigelow & Arndt, 2005; Sharma, et al. (2018).

The cognitive, emotional, and intentional contextual factors of change readiness can be expected to covary among organizational members (Weiner, 2009). As such, the relationship between generational cohorts' cognitive, emotional, and intentional behaviors toward change readiness requires further study. Additionally, Weiner (2009) suggests that healthcare systems, community health centers, and specialty medical practices hold an appeal for studies to be undertaken to test the theory of organizational readiness for change. The problem that supported this study was the lack of understanding of whether or not different positions and tenure levels impact the characteristics of generational cohorts on readiness for change. This current study seeks to understand these factors and their interaction to influence generational members' readiness for change in a healthcare climate. To further narrow the scope, the study focused on the mediated variables of position and tenure levels of organizational members to ascertain if the length of employment or the position held has any relationship to change readiness since these are contextual factors that might explain readiness for organizational change (Weiner, 2009).

# **Purpose of the research**

The purpose of this quantitative study was to determine how tenure levels and position categories influence generational cohorts' readiness for organizational change in the healthcare industry at healthcare systems across the Southeastern United States. The healthcare sector in the Southeast and the United States is undergoing change with the

emergence of Electronic Medical Records, new regulatory policies and guidelines, and the need to adapt to the increasing generational diversity of the modern workforce (Hill & Powell, 2009). The healthcare industry is becoming a precision-based industry, transitioning from population-based care to personal-based care. The healthcare service climate is challenged by the emergence of patient consumerism, and executives are under pressure to provide better, faster, and cheaper care (Hill & Powell, 2009; Himmelstein, Woolhandler, Almberg, & Fauke, 2018; Sharma, et al., 2018). As healthcare systems continue to struggle to manage these challenges, the success of the industry is dependent on the ability of managers to develop the capacity to lead these disruptions and empower employees to be ready for a change. The purpose of this quantitative exploratory study seeks to:

- a) extend the research on the readiness to change and add a generational dimension to existing studies that highlight the behavioral response to change;
- b) seek clarification of any influence that generational cohort characteristics have on the response to organizational change by change recipients; and
- c) determine if position categories and tenure moderate the relationship of generational cohorts on the readiness for organizational change among actively employed adults (≥18 years old) in healthcare organizations in the Southeastern States of the United States.

In times of change, interpersonal interaction among employees and their superiors is highly valued, making the nature of such relationships important in shaping the support for change (Bouckenooghe, De Clercq, Deprez, 2014). The present study rests on the

individual level process-of-change factors that hinge on the cognitive, emotional, and intentional dimensions that have been shown to inform employees' behavior regarding readiness for change (Bouckenooghe et al., 2009). Additionally, this study focused on four of the five current generations (Baby Boomers, Generation X, Generation Y, and Generation Z), as these groups constitute most of the modern workforce. Lancaster & Stillman (2002) argue that different generations have varying reactions and responses to organizational events because of contrasting value systems. These generational differences require organizational leadership to devise strategies which encourage readiness and commitment to change (Lancaster & Stillman, 2002). The fifth generation, the Traditionalists, born between 1925 and 1945, was not included in this study because they have reached retirement age (Tulgan, 2004).

Mowday, Porter, and Steers (1982) indicated that individuals behaviorally have a higher initial commitment to an organization, but after that engage in behaviors that enhance their performance and positional level within the workplace; tenure is most effective when categorized into career stages and examined with the age difference of employees (Wright & Bonett, 2002). This assertion indicates that there is a possible effect of an employee's commitment and readiness to engage in change initiatives as it relates to the position category or tenure level. The problem presented seeks to understand the interactive effects of tenure, job category, generational cohorts, and readiness for change. This study would clarify whether or not tenure and job category moderate the correlational relationship of generational cohorts on readiness for change. Therefore, the problem for this study was the lack of understanding of how tenure and job

category moderate the correlational relationship of generational cohorts on readiness for change.

# **Research question and hypotheses**

This study seeks to discern the interaction of tenure and job category with generational cohorts to moderate employees' readiness for organizational change. The omnibus research question is: Controlling for the moderating effects of tenure (TE) and *Position category (POC), What is the relationship between generational cohorts* (GENCO) and employee readiness for organizational change (ROC) in a healthcare *environment?* The question is rooted in the theory of generations (Mannheim, 1952; Strauss & Howe, 1991) and organizational change readiness theory (Armenakis et al., 1993; Bouckenooghe et al., 2009). Generation theory suggests that people are significantly influenced by their socio-historical environment (Mannheim, 1952). Furthermore, literature has shown that generational cohort characteristics explain the individual behavioral response to organizational change (Becton, Walker, Jones-Farmer, 2014; Van Egeren, 2009). Change readiness theory posits that people inherently resist change when and if they are not prepared for the change (Armenakis et al., 1993; Bouckenooghe et al., 2009; Kotter, 1996). The research questions, sub-question, hypotheses, sub-hypotheses were generated as follows:

# The Omnibus research question and hypothesis

**RQ0:** What is the relationship between generational cohorts (GENCO) and employee readiness for organizational change (ROC) in a healthcare environment when moderated by tenure (TE) and position category (POC)?

**H0:** There is no statistical relationship between generational cohorts and Employee readiness for organizational change in a healthcare environment when moderated by tenure and position category.

The omnibus research question implied that two questions could be derived to investigate the phenomena and require the use of correlation statistical analysis. First, the correlation statistical tool was used to assess the linear relationships among the four generational cohorts of healthcare employees and their readiness for organizational change. Second, the correlation analysis was used to estimate the relationship between the moderated variables of tenure, position category, and generation cohorts on employee readiness for organizational change. Thus, the implied questions and hypotheses from the omnibus research are as follows:

**RQ1:** What is the relationship between generational cohorts and employee readiness for organizational change?

**H**<sub>01</sub>: There is no statistically significant relationship between generational cohorts and employee readiness for organizational change.

**RQ<sub>2</sub>:** What is the relationship among tenure, position category, generational cohorts, and employee readiness for organizational change?

**H**<sub>02</sub>: There is no statistically significant relationship among the variables of tenure, position category, generational cohorts, and employee readiness for organizational change.

To answer the first question, a correlation analysis was used to investigate the relationship between the different generational characteristics of healthcare employees and their readiness for organizational change. The following are sub-questions ( $\mathbf{RQs_{1a}} - \mathbf{RQs_{1d}}$ ) and null sub-hypotheses ( $\mathbf{H_{0S1a}} - \mathbf{H_{0S1d}}$ ) for the first research question.

**RQ**<sub>S1a</sub>: What is the relationship between Baby Boomer characteristics and employee readiness for organizational change?

**H**<sub>0S1a</sub>: There is no statistically significant relationship between Baby Boomer characteristics and employee readiness for organizational change.

**RQ**<sub>S1b</sub>: What is the relationship between Generation X characteristics and employee readiness for organizational change?

**H**<sub>0S1b</sub>: There is no statistically significant relationship between Generation X characteristics and employee readiness for organizational change.

**RQ**<sub>S1c</sub>: What is the relationship between Generation Y characteristics and employee readiness for organizational change?

**H**<sub>0S1c</sub>: There is no statistically significant relationship between Generation Y characteristics and employee readiness for organizational change.

**RQ**<sub>S1d</sub>: What is the relationship between Generation Z characteristics and employee readiness for organizational change?

**H**<sub>AS1d</sub>: There is no statistically significant relationship between Generation Z characteristics and employee readiness for organizational change.

The following are sub-questions ( $RQ_{S2a} - RQ_{S2b}$ ) and sub-hypotheses ( $H_{AS2a} -$ 

HAS2b) for the second research question. The relationship of the four position categories

and tenure levels and generational cohort characteristics were correlated with the

readiness for organizational change dimensions of hospital employees. The sub-questions

and sub-hypotheses are as follows:

**RQ**<sub>52a</sub>: What is the relationship among tenure at the 0-to-2 level, 3-to-5 level, 6-to-8 level, 9+ level, generational cohorts, and employee readiness for organizational change?

**H**<sub>052a</sub>: There is no statistically significant relationship among the variables of tenure at the 0-to-2 level, 3-to-5 level, 6-to-8 level, 9+ level, generational cohorts and Employee Readiness for Organizational Change.

**RQ**<sub>S2b</sub>: What is the relationship among the position categories of medical support staff category, administrative staff category, specialty and ancillary service staff

category, data management and other position categories, generational cohorts, and employee readiness for organizational change?

**H**<sub>0S2b</sub>: There is no statistically significant relationship among the variables of position category at the medical support staff category, administrative staff category, specialty and ancillary service staff category, data management and other position categories, generational cohorts, and employee readiness for organizational change.

These sub-questions and sub-hypotheses imply the use and involvement of the correlation statistical test mentioned above. A correlation test is useful when examining the relationship between two or more normally distributed interval variables (Field, 2009; Laerd, 2013). As such, the correlation analysis was used to investigate the linear relationship between generational cohort characteristics and readiness for organizational change. Additionally, a correlation analysis was used to predict the value of one variable based on the value of one or more other variables (Field, 2009; Laerd, 2013). The correlation analysis was used to assess if there is a relational direction of the employee's position category, tenure, and generational cohort characteristics on readiness for organizational change. Per Field (2009) and Moore (2001), correlational analysis is useful and appropriate when investigating the direction and strength of the linear relationship between measurable variables.

# **Definition of terms**

Employee readiness for organizational change can be categorized into three distinct groups: intention, cognition, and emotion, and are explained by an employee's perceived understanding of the change, their attitudes towards that change, and their intended behavior towards the proposed change (Armenakis et al., 1993; Bouckenooghe et al., 2009; Holt et al., 2007). This study focuses on the difference of generational

cohorts, their readiness for change, and the impact of tenure and position category levels. The following definitions, operational definitions, and acronyms are used in this study.

*Ambivalence:* The negative and positive attitude manifested as a reaction towards an event (Oreg & Sverdlik, 2011).

*Baby Boomers:* Individuals born between the years 1946 and 1964 (Dimock, 2019; White, 2006; Jefferies & Hunte, 2004).

*Continuous change:* A constant, gradually developing, and increasing change without a defined end state which involves freezing, rebalancing, and refreezing (Szabla, 2007; Weick & Quinn, 1999).

*Episodic change:* The intentional changes that are infrequent, deliberate, irregular and involve the concept of unfreezing, transitioning, and refreezing (Lewin, 1947; Weick & Quinn, 1999).

*Employee readiness for organizational change (ROC):* The attitudes, beliefs, and intentions of employees which predict behaviors such as resistance to, or support of, organizational change initiatives (Armenakis et al.,1993; Holt et al., 2007).

*Generation X:* Individuals born between the years 1965 and 1980 (Dimock, 2019; White, 2006; Jefferies & Hunte, 2004).

*Generation Y:* Individuals born between the years 1981 and 1996 (Dimock, 2019; White, 2006; Jefferies & Hunte, 2004).

*Generation Z:* Individuals born between the years 1997 to the present. A chronological endpoint has not been set for this age group. However, this study follows

the age range proposed by the pew research center (Dimock, 2019; White, 2006; Jefferies & Hunte, 2004).

Generational Cohort: "A group whose length approximates

the span of a phase of life and whose boundaries are fixed by peer personality." (Strauss & Howe, 1992, p.60)

*Generations:* A group of individuals who share a range of birth years, personality traits, and life experiences. (Lancaster & Stillman, 2002).

*Organizational Change (OC):* The planned and unplanned process of altering the way things are done in organizations (Van de Ven & Poole, 1995)

*Readiness for change:* See also Employee Readiness for Change.

In addition to "Readiness for Change" and Employee Readiness for Change, the

terms "Change Readiness," "Employee Readiness," and "Employee Readiness for

Organizational Change" are used interchangeably throughout this study. Further, the

terms Generations and Generational Cohorts are also used interchangeably. Finally,

Generation Z and iGeneration (iGen) are also interchangeable terms.

# **Operational definitions**

The following are operational definitions for readiness for change, generational cohorts, tenure, and position categories.

# The operational definitions of the employee Readiness for Organizational Change (ROC) variables are:

- a) *Intentional readiness for change*: This is the extent to which employees are prepared to exert energy to effectuate the change process.
- b) *Cognitive readiness for change*: This is composed of the inherent beliefs and thoughts employees hold regarding the change to be undertaken.

c) *Emotional readiness for change*: This captures the feelings or emotional states of the employees toward change. (Armenakis et al., 1993; Bouckenooghe et al., 2009; Holt et al., 2007).

# The operational definitions of the generational cohort variables are (GENCO):

- a) *Baby Boomers*: The Baby Boomer generation includes individuals born between the years 1946 to 1964, age 55 to 73;
- b) *Generation X*: Generation X is composed of individuals born between the years 1965 to 1980, age 39 to 54;
- c) *Generation Y*: The Millennial generation include individuals born between the years 1981 to 1996 age 23 to 38;
- d) *Generation Z*: The iGen or Centennial generation is composed of individuals born between the years 1997 to later, age 18 to 22. (Dimock, 2019; White, 2006; Jefferies & Hunte, 2004).

# The operational definitions of the Position Categories and Tenure variables are:

- a) *Position Categories (POC)*: Area in which individuals perform job functions
  - i. Medical Support Staff Doctors, Nurses, and Technicians
  - ii. Administrative Staff Executives, Managers, Supervisors, Finance,
    Human Resources, and organization operations staff
  - iii. Specialty and Ancillary Service Staff Laboratory, Cardiology,

Customer Service, Housekeeping, food services, and other lay staff.

- iv. Data Management and Other Staff IT, HIM, and other organizational personnel not mentioned above.
- b) *Tenure (TE)*: Number of years worked at the Organization
  - i. 0-to-2 years
  - ii. 3-to-5 years
  - iii. 6-to-8 years
  - iv. 9+ years

## Acronyms

- BABO Baby boomer generation
- CRC Cognitive readiness for change
- ERC Emotional readiness for change

GENCO - Generational cohorts

GENX – Generation X

GENY – Generation Y

iGEN – Generation Z

IRC – Intentional readiness for change

- POC Position category
- ROC Employee readiness for change

TE - Tenure

- DV Dependent variable
- IV Independent variable
- MV Mediating variable

## Assumptions, Limitations, and Delimitations

This study rested on assumptions that are theoretical, topical, and methodical:

**Theoretical assumption.** Per generation theory, generational cohorts identify specific behaviors, feelings, and thoughts that are formative to experiences (Mannheim 1952). The concept of employee readiness for change theory maintains that resistance to change occurs when employees are not ready for a change. This study examines the relationship between generational cohorts and employee readiness for organizational

change. The assumption is made that the theories that guide this research are appropriate for the study.

**Topical assumption.** The current literature on change readiness suggests that employees must be ready for the change if change efforts are to be successful. The lack of readiness for change is often manifested in characteristic behaviors such as resistance, cynicism or ambivalence (Armenakis et al., 1993). Further, Bouckenooghe et al. (2009) assert that there is a high cost associated with the behavior of employees toward change initiatives. As such, there is continued interest by organizations to understand the dynamics of organizational change. With increasing generational diversity in the workforce, it is assumed that organizations would want to know whether or not generational cohorts inform the behavior of employees toward change initiatives, and whether or not job categories and tenure improve or discourage employee readiness for organizational change. It is assumed that this study will expand the understanding of organizations on whether or not generational differences impact the readiness for change in employees. It is also assumed that healthcare leaders would be able to assess their organization's readiness for change with the information presented in this study.

**Methodological assumptions.** This study will investigate the relationship among generational cohorts and their readiness for change. The study will also examine the direction and strength of the linear relationship of tenure and position categories of generational cohorts on employee readiness for change. Since a correlational analysis is applied to this study, it is assumed that the correlation analysis procedures that will be used in this study are appropriate because correlation is useful for examining the relationship of two or more variables for statistical significance and is valid for observing

linear relationship between the variables by analyzing the correlation coefficients (Field, 2009; Laerd, 2013; Moore, 2001). It is also assumed that the samples were representative of the healthcare population and inference can be made about the change readiness of employees, that the instruments of measurement were appropriate for capturing necessary information, and that the study participants were honest in their responses.

Limitations and delimitations. There are limitations associated with the use of surveys to gather data from large populations even though they are an appropriate methodology for research. As a limitation, the data collected are only accurate to the extent that study participants are honest and understand the questions when reporting their answers. Since the healthcare sector—particularly the hospital and medical facility environment—is vibrant and dynamic, it may be challenging to obtain responses from all positions of healthcare employees, and the results may not be representative of all healthcare environments. Nevertheless, the study is delimited to healthcare systems employees within the Southeastern United States to narrow the scope of research.

### The significance of the study

Organizations are becoming global and are faced with the continual need for change. They require adaptive employees who are receptive to change initiatives to ensure success. The study of change within organizations is essential to the practice of management and defines organizational behavior (Kitchen & Daly, 2002). Therefore, this study significantly contributes to the field of study on organizational change management and change strategies. Although not global, in the United States the healthcare sector is undergoing a necessary and tremendous amount of change, and is becoming a precisionbased industry by moving away from population-based care to personal-based care. There

is little research on change readiness of generational cohorts in the healthcare industry. The healthcare industry is struggling to manage these changes, and the success of the industry is dependent on organizational leaders' ability to develop the capacity to lead the disruption to "normal" business practices. First, this study will equip healthcare administrators with knowledge of how different generations respond to change initiatives. Second, the healthcare sector will be better equipped to implement change initiatives successfully. Third, this study will extend the literature on organizational change and reveal the role that generational cohorts play in employee change readiness when moderated by tenure and position descriptions.

## The theoretical and conceptual framework

Two theoretical frameworks govern this research: change readiness theory (Armenakis et al., 1993; Bouckenooghe et al., 2009) and generational theory (Mannheim, 1952; Strauss & Howe, 1991). The generational cohorts, moderated by the position category and tenure of employees, is used to predict the readiness for change (Figure 1).



# Figure 1: Conceptual Framework: Generational Cohort on Employee Readiness for Organizational Change Moderated by Tenure Level and Position Category.

This study will explore whether or not (1) the characteristics of the different generational groups have any relationship to employee readiness for organizational change, and (2) the generational cohorts on employee readiness for organizational change are moderated by position categories and tenure. The conceptual framework of change readiness suggests that employee attitude towards change is multi-faceted and is categorized into cognitive, emotional, and intentional dimensions (Bouckenooghe et al., 2009; Armenakis et al., 1993). These dimensions of employee attitudes inform the readiness for organizational change.

# **Researcher's perspective**

The researcher experienced resistance to change when contracted to consult with a governmental healthcare organization in Atlanta, Georgia. Witnessing the various reactions of resistance and acceptance towards proposed changes, the researcher became interested in studying how different generations react to change initiatives. From this experience, the researcher believes the resistance and acceptance behaviors witnessed were due to the lack of knowledge about the proposed change and the lack of readiness for the change. Further, the healthcare sector was chosen as the focus of this study because the researcher worked in the healthcare industry for many years and has always been interested in how the sector impacts many lives. The researcher also experienced the impact of organization change as a cancer patient in the late 1990s and early 2000s. Thus, the researcher believes that when the healthcare industry is better prepared for change,

better care will be provided to the general population. The researcher also believes the Baby Boomer generation is more likely to resist change as opposed to the other generations.

# Summary

Chapter one identified the problem which inspired this dissertation effort and introduced the questions that governed and guided this research. The chapter also presented the theoretical and conceptual framework which directed this study. Further, the objectives of the study were outlined, and key terms defined. The remainder of this dissertation is arranged as follows: (a) Chapter two presents a review of extant literature and the conceptual framework that informed this research, (b) Chapter three describes the research methodology, (c) Chapter 4 presents the analysis and interpretation of the findings, (d) Finally, Chapter five introduces the conclusions about the research findings and provides recommendations for future research.

# **CHAPTER 2- LITERATURE REVIEW**

The literature on organizational change is extensive, omnipresent, and has been shown to be disruptive when planned or unplanned change is introduced in continuous, episodic, incremental, transformative, or radical manner (Axley & McMahon, 2006; Becker, et al., 2005; Burnes & Jackson, 2011; Choi & Ruona, 2011; Jacobs, Witteloostuijn, & Christe-Zeyse, 2013; Poole, 1998; Vakola & Nikolaou, 2005). Organizational change is a risky but essential strategy and proposition which often results in low success rate, with a change initiative failure rate upwards of 70% (Burnes & Jackson, 2011; Jacobs, et al., 2013). This study is built on the premise that change initiatives fail because employees are not ready for change, and that organizations can gain understanding as to why change initiatives fail. This study is also established on the premise that the differences among generational cohorts might inform change readiness and significantly contribute to the narrative and literature on organizational change.

For this research, a holistic approach is used to investigate the topic of change and its impact on individuals and organizations. Therefore, a discussion about the historical view of organizational change, readiness for organizational change, resistance to organizational change, and the cost and failures of organizational change will be presented, followed by discussions on generational cohorts, the historical view of generation theory, and generation theory in relation to readiness for organizational change. This chapter will also discuss tenure and position category in relation to employee readiness for organizational change. The purpose of this chapter is to review the literature on the foundational concepts of this research: Generational cohorts and readiness for organizational change.

### Historical view of organizational change

The literature on change and organizational change is vast and continues to present an ongoing challenge for management scholars and practitioners as these extant literatures sought to identify typologies that fail to adequately define the concept of change (Suddaby & Foster, 2017). The literature on change and organizational change extends for over a century and covers the seminal works of pioneers such as Frederick Taylor (1856 - 1915), whose book Scientific Management (c.1911) elucidated on the importance and means of managing change scientifically; the researches performed by Lewin (1947), Coch and French (1948), Grenier (1967), Yuchtman and Seashore (1967), Weick (1969), and Tushman and O'Reilly (1996) all highlighted the intricacies of change theory and contributed to the communication to define change in the life of organizations. Organizational change is an on-going process in the life of organizations and is identified as a change from one state to another, driven by forces both internal and external, planned or unplanned, continuous or episodic, and influenced by political, social, and environmental demands (Axley & McMahon, 2006; Barnett & Carroll, 1995; Becker et al., 2005; Choi & Ruona, 2011; Karp & Helgø, 2008; Suddaby & Foster, 2017; Vakola & Nikolaou, 2005; Van de Ven & Sun, 2011; Weick & Quinn, 1999).

Axley and McMahon (2006) and Schweiger, Stouten, and Bleijenbergh (2018) suggest that organizations are challenged to adapt and respond to a continuous volatile change environment quickly. However, the high cost and failure rate of change initiatives demand the effective management of organizational change and continue to drive organization change research. Effective organizational change management continues to be of great importance and a strategic focus of organizational change management studies
(Bouckenooghe et al., 2009; Oreg et al., 2011; Oreg, Bartunek, Lee, & Do, 2018). As such, the assertion is made in some extant literature that individual-level elements continue to be critical to understanding the intricacies of organizational change and identify the dynamics of behavioral and psychological reactions to change (Oreg, et al., 2018; Suddaby & Foster, 2017; Yang, Choi, & Lee, 2018). Employee reaction to and readiness for change remain an area requiring further research to understand the behavioral and psychological dynamics of organizational change (Bouckenooghe et al., 2009; Yang, et al., 2018).

Organizational change from the 1950s – 1960s: The topic of organizational change has been a research topic since the seminal work of Kurt Lewin in 1947. Theoretical experts such as Kotter, Armenakis, Holt, Harris, Field, and Bouckenooghe are among the many researchers who significantly contributed to the field of organizational change (Armenakis, & Bedeian, 1999). Between 1950 and 1960, organizational change emerged from the principles of organizational development and greatly focused on the change agents' ability and proficiency to analyze core organizational relationships (Sanzgiri & Gottlieb, 1992). Sanzgiri and Gottlieb (1992) further assert that the evolution of organizational development highlights the elements of effective communication within organizations as well as the impact of behavioral patterns of individual members of the organization, and objective procedures were developed. Here, organizational development enabled leaders to gain knowledge and problemsolving abilities during organizational change efforts, allowing leaders to develop effective strategies for change efforts (Sanzgiri & Gottlieb, 1992).

**Organizational change from the 1970s – 1980s:** The 1970s and 1980s brought an emphasis on the cultural, social, and political context of organizational change (Burnes, 2005; Sanzgiri & Gottlieb, 1992). Sanzgiri and Gottlieb (1992) argue that the economic challenges of the 1970s allowed for organizations to experience significant change where there was considerable emphasis on measurable results. Here, the conceptual principles of organizational development evolved from long-term change strategies to more systematic, measurable, short-term strategies which included employee involvement in organizational change efforts. Sanzgiri and Gottlieb (1992) and Burnes (2005) intimate that there was a systematic view of organizational development in the 1980s which resulted in the focus on the corporate culture within organizations. The cultural context of change promoted the principles of co-operative change which promote cultural excellence by creating an environment where flexibility towards culture is believed to promote innovation (Burnes, 2005). Burnes (2005) further argued that the cultural context of organizational change later emphasized the importance of the social context of change, which later highlighted the role organizational politics play during organizational change initiatives.

Organizational change from the 1990s – 2000s: Organizational change researchers from the 1990s and 2000s studied the prevalent contextual, process, climate, valence, and efficacy of change within organizations. Armenakis and Bedeian (1999) and Weiner (2009) described themes in organizational change research which describe its content, valence, contextual, process, informational assessment, and criterion factors. Studies conducted surrounding content emphasize the need to understand contemporary changes; contextual studies assess the internal and external environmental conditions of

organizations such as culture, resources, and governmental policies; the process studies highlight the implementation activities of proposed change initiatives, and the criterion studies assess the result of change initiatives (Armenakis & Bedeian, 1999). The change valence studies describe the construct which allows researchers to identify the theoretical drivers which impact the individual readiness for organizational change (Weiner, 2009). Weiner (2009) attests that studies on information assessment describe resource perceptions and situational factors that impact the readiness for change. Moreover, the 1990s and 2000s also brought forth an emphasis on change resistance and change readiness, focusing on elements such as change commitment and efficacy, and changerelated efforts, focusing on elements such as change initiation, persistence, and cooperative behavior (Armenakis & Bedeian, 1999; Armenakis et al., 1993; Bouckenooghe et al., 2009; Weiner, 2009). The increased research in the 20<sup>th</sup> and 21<sup>st</sup> century, however, expanded the understanding on the topic of change management but failed to mitigate the high costs and failures of organizational change (Burnes & Jackson, 2011; Jacobs, et al., 2013).

### The cost and failures of organizational change

Armenakis et al. (1993) and Stouten, et al. (2018) argue that organizational change efforts are costly for businesses and often result in failures because of the struggle to create successful and sustainable change which impacts their viability. A consensus of several extant literatures intimated that more than 70% of organizational change initiatives fail and result in the loss of a significant amount of dollars in operation capital; and the assertion is made that this failure and cost is attributed to the lack of employee readiness and resistance to the initiated changes (Burke, 2013; Burnes 2004; Higgs &

Rowland, 2005; Kotter, 1996; Miller, 2002; Pellettiere, 2006; Bateh, Casteneda, & Farah, 2013; Strebel, 2009; Warrick, 2009). However, Hughes (2011) argues that the universal 70% consensus touted by many researchers is an unscientific and estimated figure with a proposed range of 50% to 70% in the early 1990s by organizational change researchers, Hammer and Champy (1993).

Nevertheless, Cândido and Santos (2015) provided specific and quantifiable examples of organizational change failures. For example, Cândido and Santos (2015) asserts that 30% of change projects have been abandoned, organizational change initiatives such as joint ventures often result in a 61% failure rate, technological and manufacturing change projects possess a failure rate of 81%, and 91% of total quality management (TQM) change programs fail (Cândido and Santos, 2015). Burnes (2004) concludes, as a result of a literature review, that change failure rate, as a product of organizational change resistance, is higher than 70%. Decker, et al. (2012) confirmed this argument with the assertion that the rate of change failure is much closer to 93%.

Regardless of the rate at which organizational change fails, several extant studies have concluded that the reason for the failure of change efforts is attributed to the lack of understanding about the role that employees play in organizational change initiatives (Armenakis et al., 1993; Bouckenooghe, 2009; Holt et al., 2007). Employees resist change even if the initiated change serves their interests (Bouckenooghe, 2010; Oreg, 2003), resulting in high failure rates and significant cost for organizations. This current high failure rates of organizational change programs coupled with the associated high costs suggest that the topic of organizational change readiness remains an area of further study and interest.

#### **Readiness for organizational change**

Extant literature on organizational change suggests that a change from one state to another is driven by planned and unplanned internal and external forces, and factors such as competitive pressures, technological innovations, mergers and acquisitions, changing consumer tastes, political, social, and environmental demands exert significant influence on organizations to implement change (Axley & McMahon, 2006; Barnett & Carroll, 1995; Becker et al., 2005; Choi & Ruona, 2011; Karp & Helgø, 2008; Vakola & Nikolaou, 2005). Organizational change has been labeled in various ways throughout history, including terms such as "organizational development," "downsizing," "restructuring," "re-engineering," "outsourcing," and "technological advancement" (Kotter, 2007). Organizational change efforts can coincide and involve several types of change, and regardless of the influencing factors, organizational change can be episodic and continual (Bommer et al., 2005; Weick & Quinn, 1999; Smith, 2002). For example, change initiatives can involve significant organizational modifications such as mergers and acquisitions, or incremental changes that affect the overall company processes (Burke, 2013).

The concept of readiness for change theory indicates that employees are more likely to support change initiatives when they are expecting and prepared for the change (Armenakis et al., 1993; Bouckenooghe et al., 2009). As such, the conceptual framework of change readiness suggests that employee attitudes toward change are multi-faceted and are categorized into cognitive, emotional, and intentional dimensions (Bouckenooghe et al., 2009). These dimensions of change readiness (cognitive, emotion, and intention) are behaviorally expressed as ambivalence; support of, or resistance to organizational change

(Armenakis et al. 1993; Holt et al., 2007). Szabla (2005) states that resistance to change is best understood from the cognitive, emotional, and intentional dimensions. This assertion indicates that resistance to change is informed by the readiness for change.

Further, Bouckenooghe et al. (2009) measured the three dimensions of readiness for change outlined in the Employee Readiness for Change section of their Organizational Change instrument, and this conceptualization will guide this research as the instrument of choice to capture data on readiness for organizational change. Since organizations continue to be interested in change management because of unsuccessful change initiatives and high failure rate, and with increasing generational diversity of the workplace and unprecedented change occurring in the healthcare sector, understanding how generational characteristics impact readiness for change might enable healthcare leaders to understand if generational differences might be the reason for the high failure rate of change programs (Bouckenooghe et al., 2009). Further, Szabla (2005) asserts that since resistance to change is informed by readiness for change, organizational leadership must understand the three dimensions of cognition, emotion, and intention.

The cognitive dimension of change readiness. The cognitive dimension of organizational change readiness has been highlighted as an attitude or behavior manifested as a precursor of resistance or support of change initiatives (Armenakis et al., 1993; Lamm & Gordon, 2010; Oreg & Sverdlik, 2011). The thought process of individuals about change initiatives is significantly critical to the outcome of change programs (Kitchen and Daly, 2002). Here, the assertion is made that how employees think about change initiatives is more important than how they act towards change. The cognitive dimension focuses on what employees think and understand about change, a

process in which employees appraise, perceive, or understand the change (Armenakis et al., 1993). Piderit (2000) states that the cognition responses of employees inform their decision to support or resist organizational change. As such, cognition is a state of mind which reflects the receptivity of employees to embrace initiated changes; it is a perception of the need for change and the readiness of the organization to successfully employ the change (Armenakis & Harris, 2002). The cognitive dimension of readiness for organizational change has been the focus of many types of research on readiness for organizational change because of its impact on the success or failure of change initiatives. As such, there is a possibility that cognition, which speaks of rationalization, might influence how generational employees respond to support or not support change efforts.

The emotional/affective dimension of change readiness. Emotional or affective dimensions of change readiness correspond to positive, negative, neutral, or mixed emotional behaviors relating to organizational change (Armenakis et al., 1993; Lamm & Gordon, 2010; Oreg & Sverdlik, 2011. Here, positive affective behavior is seen as a support of change initiatives, negative affective behavior as resistance to organizational change, neutral affective behavior reflects neither support for nor resistance to change, and the mixed affective behavior reflects partly positive and partly negative reactions about initiated changes. According to Clore, Wyer, Dienes, Gasper, Gohm, and Isbell (2001), emotion or affect influences individual perception positively or negatively, impacting how employees respond to organizational change efforts. Further, a positive affective dimension enables employees to rely on established knowledge structures to make decisions about the initiated change while the negative affective dimension negates reliance on existing knowledge structures to make decisions about change programs

(Bless, 2001). Positive affect and negative affect dimensions have been heuristically labeled as influencers of employee behavior where positive emotion is useful in directing employee attention to new information, and negative emotion influences behavior through avoidance (Bless, 2001; Hoffman, 1986; Staw & Barsade, 1993). According to Bouckenooghe (2010), change readiness is a positive attitude toward change and explains the values and desires of employees when new changes are proposed.

The intentional dimension of change readiness. The intentional dimension of change readiness describes the propensity or willingness of employees to exert their energy to engage in the change process (Oreg, 2003). Per Piderit (2000) the intentional dimension highlights the attitudes of the employee and informs behavior. Intentional dimensions influence the attitude and disposition of employees, predicting behaviors such as ambivalence toward, support of, or resistance to change initiatives (Armenakis et al., 1993; Lamm & Gordon, 2010; Oreg & Sverdlik, 2011). Intention, along with emotion and cognition, provides a framework in which feelings or attitudes about change initiatives are interconnected with the thought processes and behavioral emotions about the proposed change programs (Oreg, 2003). Because intention is associated with emotion and cognition, this study will evaluate this framework with a generational cohort perspective to investigate and understand readiness for organizational change.

# **Resistance to organizational change**

The resistance to organizational change is a significant factor that impacts the effectiveness of change efforts, is comprised of three dimensions: emotional, intentional, and cognitive, and has been found to be contradictory in definition in extant literature (Akan, Er Ülker, & Ünsar, 2016; Chung, Su, & Su, 2012; Grama & Todericiu, 2016;

Schweiger et al., 2018). Lewin (1947) conceptualized and categorized the change process as unfreezing, moving, and refreezing as a contextual change behavior (Lewin, 1947). The concept of resistance to change was first popularized by Coch and French (1948) in the paper "Overcoming Resistance to Change." The organizational context presented by Lewin (1947) and Coch and French (1948) considered individual employees as factors to successful change initiatives. Burnes (2015) posits that the concept of change resistance shifted from a focus on an organizational perspective to a personal perspective, which highlights the psychological behavior of individuals. As such, resistance to organizational change can be viewed from individual responses and motivations based on the desire for the proposed change or lack thereof, the lack of understanding the proposed change, the belief that the change will produce negative results, and the lack of tolerance for the proposed change (Kotter & Schlesinger, 2008).

#### **Employee response to organizational change**

Employee response to change is often recorded as a resistance to organizational change due to the uncertainty and fear of the impact of the change, the misalignment with the personal priorities of employees, the inadequate communication about the details of the change between the change agents and employees, and lack of resources to implement the change (Adcroft et al., 2008; Armenakis et al., 1993; Zwick, 2002; Oreg, 2003; Oreg et al., 2011). Change initiatives are most successful when employees respond in support of the initiated change (Armenakis et al., 1993). When employees respond to proposed changes by embracing the status quo and previously held beliefs, however, organizational change objectives can become stagnant, requiring a process of unfreezing through the development of messages to efficiently communicate the importance and significance of

change initiatives (by Armenakis et al., 1993; Lewin, 1947). Moreover, when employees respond to change in the manner described above, it can be explained in the three dimensions of change readiness: cognition, emotion, and intention (Armenakis et al., 1993; Bouckenooghe et al., 2009; Oreg et al., 2003; Oreg, et al., 2018). Finally, some extant literature suggests that employees' change resistance can be alleviated when employees are prepared (Armenakis et al., 1993; Bouckenooghe et al., 2009; Lewin, 1947).

# Positive view of change resistance

Individuals who respond to organizational change positively are valued by organizations (Oreg, 2003). However, Oreg (2003) further proclaims that employees resist change because of fear, uncertainty, misalignment of interest, poor communication, and lack of resources. The concept of change resistance is thus viewed negatively as a result of the high cost associated with change failures and is deemed to be a resistance to organizational change initiatives by employees (Armenakis et al., 1993; Oreg, 2003). Nevertheless, not all change resistance negatively impacts organizations. Resistance is viewed positively because of the value it can produce for organizations; it sometimes results in the prevention of potentially detrimental change from being initiated (Bareil, 2013; Ford & Ford, 2009; Zwick, 2002). Avey, Wernsing, and Luthans (2008) assert that the positive view of change resistance is worth further consideration and study. However, the focus of this research is to assess the lack of readiness of the generational cohort employees for organizational change and identify the characteristics which contribute to resistance to change and the high cost of change failures.

# Summary

The high failure rate of organizational change initiatives continues to present problems within businesses across the United States, and as such, there continues to be increased interest in the study of organizational change (Bouckenooghe et al., 2009). The increasing generational diversity of the modern workforce continues to challenge healthcare leaders as they navigate their many change initiatives (Lancaster & Stillman, 2002; Sharma, et al., 2018). Understanding the change phenomena and the role generational employees play in the readiness for change might clarify the reasons behind the continually high rate of failure in change initiatives (Becton, 2014; Bouckenooghe et al., 2009; Bourne, 2015). Since position and tenure are factors which can influence change, and there is a moderating effect of employee tenure and position groups on organizational commitment and outcomes (Cohen, 1991, 1992; Weiner, 2009), understanding the relationship between different generational groups with different position categories and tenure levels might help organizational leaders identify the relationship between generational cohort behavioral characteristics and the readiness for change.

#### **Generational Cohorts Theory**

As reported by Lewis (2006), current literature on change management suggests that individual cooperation or resistance throughout organizational change initiatives is essential to achieving success or failure. Change management theory and generational theory are popular subjects of examination in the extant literature. However, the phenomenon of how generations respond to change initiatives regarding resistance needs further exploration. The increasing generational diversity in the workplace has led to four

generations with different characteristics and life experiences working within the same organizations (Lancaster & Stillman, 2002; Jeffries & Hunte, 2003, White, 2006). It is essential to understand the impact of these generations on change programs and determine their effect, if any, on projecting the success of change initiatives (Lewis, 2006). The focus of this literature review is to evaluate the generational perspective on change initiatives by focusing on the following: the emerging multigenerational workforce, historical view of generation theory, generational theory, generational differences and similarities, and generational response to organizational change.

**Multigenerational workforce.** Multigenerational employees with diverse backgrounds demonstrate various characteristics, preferred communication styles, values, and career outlooks; these preferences and attributes are referred to as the clash point of generation gaps (Zemke, Raines, & Filipczak, 2000). Understanding the motivations, strengths, weakness, and actions of each generational group is deemed a necessary fixture in determining the readiness for organizational change of each cohort. As specified by Martin and Tulgan (2006), it is incumbent upon organizational leadership to learn, adequately assess, and captivate each generational cohort in the workplace, implementing programs that correspond to the characteristics and style of the different generations. It is necessary, therefore, for organizational leadership to ascertain the benefits of the multigenerational diversity of the modern workforce and harness its potential to increase the success of change programs.

# Historical view of generation theory

Generational research seeks to understand the relationship of various generations relative to the field of interest for a researcher. Differences in some generational research

exist when categorizing cohort definitions and characteristics in some extant literature (Tulgan, 2009; Twenge, 2006). Becker (1960) argues that generational cohort research has been a topic of discussion since the 1960. However, there is evidence in the extant literature supporting generational research before the 1960s (Mannheim, 1952) and popularized in the 1990s by Strauss and Howe (1991). The historical view of generation theory is a concept informed from the theory of social identity (Stets & Burke, 2000). Although there is a foundation of research in some extant literature with agreement on the differences of various generational cohorts today (Strauss & Howe, 1991, 1992; Tulgan, 2009; Twenge, 2006; Zemke al., 2000), the seminal works of Mannheim (1952) and Strauss and Howe (1991) provide the historical basis for all future research.

**Mannheim theory of generations.** The seminal work of Karl Mannheim (1952), a sociologist, explored the concept of generations from a sociological perspective by focusing on the historical and life experiences of various generational cohorts. Mannheim (1952) argued that social and cultural factors are significant elements of consideration when evaluating generational differences rather than biological factors. Here, the work of Mannheim (1952) suggests that the social and cultural factors experienced by different generations can be used to validate the characteristics of the generations by categorizing common behavior patterns. Per Corsten (1999) and Foster (2013), the theory proposed by Mannheim focuses on the cognitive and emotional attitudes and actions experienced by generations during their lifespan. Further, the argument is made that each generational cohort possesses a unique sociological paradigm when evaluating organizational goals and objectives as well as life experiences (Mannheim, 1952). As such, Mannheim's

theory of generations suggests that different generational cohorts could react differently to organizational goals and objectives because of differing viewpoints and perspectives.

Strauss and Howe generational theory. The research of William Strauss and Neil Howe (1992) was the first to present the concept of generational differences, and described generational units as a subgroup within a generation time span. The generational theory proposed by Strauss and Howe (1992) maintains that peer personalities define generational groups and that generations are categorized into cohorts with the length of each group approximating the life span of the cohort group. According to Strauss and Howe (1992), generational cycles have a historical basis and forecast the direction and category of future generations. Further, Strauss and Howe (1992), argued that social and environmental events impact generations. Here, their research demonstrated the characteristics of each generational cohort, identified the historical evidence to affirm the cyclical movements of each generation, and integrated identified generational characteristics with research performed on the bases of birth rates, sociocultural trends, and environmental events impacting populations. This contribution from Strauss and Howe (1992) served as a framework for future studies with the proposed principles employed as a foundation in generational research within organizations.

# **Description of Generations**

Organizations expend significant resources to understand and analyze the life stages, ethnicity, religion, and behavioral styles of their employees, but little emphasis is placed on the diversity of generational differences (Lancaster & Stillman, 2002). The significance of increasing generational differences requires that organization leaders gain

an understanding of these differences and distinguish how generational differences impact change resistance in the workplace (Sessa, Kabacoff, Deal, & Brown, 2007). As mentioned by Lancaster & Stillman (2002), generational differences can promote conflict, reduce productivity, increase stress, and increase employee turnover, making it significantly difficult for change initiatives to be successful. Generational cohort theory presents the premise that the values practiced by various generational cohorts are based on the social norms and behavioral values developed by each generation (Blythe et al., 2008). Thus, it is expedient to evaluate the generational impact on change readiness. Nevertheless, little quantitative research on generational cohorts exists (Strauss & Howe, 1992).

Mannheim's (1952) seminal work on the theory of generations describes how life events shaped by the experiences and worldviews of people across class, racial, and geographic boundaries tend to have similar thought processes, reactions, and behaviors. As stated by Strauss and Howe (1992), generational cohorts "…encounter the same national events, moods, and trends at similar ages. They retain, in other words, a common age location in history throughout their lives" (p. 48). However, managing the diversities of generations of employees is a leadership challenge when implementing change initiatives. Per Jeffries and Hunte (2003), all generation cohorts have disparate value systems and respond differently to situations, and the knowledge of generational characteristics provides an understanding of the diversified workforce as well as the personal motivators of employees; navigation of the diversified workforce presents a leadership challenge for organizational leadership.

**Baby boomer generation:** The first generation for discussion is the Baby Boomers. This generation is known by various names such as the Woodstock, Love, and Me generation (Dahlroth, 2008). The Baby Boomer generational cohort includes individuals born between the years 1946 and 1964 (Dimock, 2019; White, 2006; Jefferies & Hunte, 2004). Members of this cohort were born to Salient or Traditionalist generation parents who experienced World War II, the Korean War, and the Great Depression (Lancaster & Stillman, 2002; Seung-Bum & Guy, 2006). As stated by Nelson (2007), the Baby Boomer generation presents the largest generational cohort of employees in the modern workforce, accounts for 52% of the workforce, and currently holds middle and executive level management positions within organizations. Moreover, research by Farag, Tullai-McGuinness, and Anthony (2009) indicates that the Baby Boomer generations' composition of the modern workforce is decreasing as this generation makes up 47% of the workforce. Some research on generational cohorts contend that the Baby Boomer generation represents approximately 80 million people in the United States (Dimock, 2019; Hobbs & Stoops, 2000; Lancaster & Stillman, 2002; Strauss & Howe, 1992).

The Baby Boomers have been categorized stereotypically as being highly focused on attaining achievements, independent, assume control of their fortunes, respect authority, and maintain a steady and consistent disposition in the work environment (Becton, et al., 2014). In the work environment, Baby Boomers are considered to be highly competitive, tend to measure their success materially, are self-reliant, and maintain a centrality of work or are considered workaholics (Becton, et al., 2014). Further, Baby Boomers demonstrate dedication and loyalty as values in the workplace. However, Parry (2017) attests that the Baby Boomer generation views career and life as one and the same,

and as a result, Baby Boomers are the most stressed generation because of their commitment to work and the desire to succeed in life. Moreover, these qualities of identification for the Baby Boomers are different for proceeding Generation X, Y, and Z.

Generation X: The second generation for discussion is Generation X. This group is also identified by names such as GenX, GenXers, Latchkey Generation, and the Lost Generation (Strauss & Howe, 1991). Generation X generational cohorts are individuals born between the years 1965 and 1980 (Dimock, 2019; White, 2006; Jefferies & Hunte, 2004). Members of the GenX generational cohort represent approximately 46 million employees in the modern workforce (Lancaster & Stillman, 2002), are employed in lower-paying jobs (Jeffries & Hunte, 2004), and compose of the smallest cohort with social and economic experiences that are different from the Baby Boomers (Strauss and Howe, 1991,1992). The Generation X cohort is characterized by events such as the Vietnam War, the oil and energy crisis of the 1970s, the fall of the Berlin Wall, the Cold War, and the economic inflation and uncertainties of the 1980s (Dimock, 2019; Lancaster & Stillman, 2002; Strauss and Howe, 1992).

Per Lancaster and Stillman (2002), Generation Xers are misunderstood in the workforce and often seen by the Baby Boomer generation as slackers; however, this group has its unique identity and is an influential population in the modern workforce. In the workplace, members of Generation X are characterized by their independence, are highly family focused, resilient, critical of changes and other generational groups, adaptable, and hardworking; however, the Generation X cohorts tend to be intolerant of workplace bureaucracy and are socially responsible (Johnson & Johnson, 2010; Murphy, 2007). According to Hill (2004), Generation Xers prefer flexible work schedules in an

informal work environment and are comfortable with change as long as a work-life balance exists, and the quality of the result is valued over quantity. The GenX cohort members are quick learners who seek the attainment of new knowledge and skillsets, embrace diversity in the work environment, and are more comfortable with change because of their experience of change growing up (Aldisert, 2002; Bursch & Kelly, 2014; Holtshouse, 2010).

Generation Y: The next generation for discussion is Generation Y. Members of this generational cohort are also known as the Millennials, Gen Yers, Echo Boom, Nexter Generation, and the Net Generation (Dimock, 2019; Horovitz, 2012; Lancaster & Stillman, 2002). The Millennial generation includes individuals born between the years 1981 to 1996 (Dimock, 2019; White, 2006; Jefferies & Hunte, 2004). Members of the Generation Y cohort are shaped by experiences such as 24/7, unlimited access to the Internet; personal cell phones and digital cameras; and historical events such as various terrorist attacks, the end of the Cold War, the Oklahoma City bombing, Operation Desert Storm, and the emergence of social networks (Murphy, 2007). Generation Y is projected to be include more than 81 million in the U.S., approximately one-fourth of the population of the United States (Bursch & Kelly, 2014; Rawlins, Induik, & Johnson, 2008). GenY and has had more exposure to modern technological advances since their childhood than previous Baby Boomer and GenX generations (Lancaster & Stillman, 2002). According to Bursch & Kelly (2014), as of 2014, the Millennials generation composed 36% of the workforce, and is projected to represent approximately 46% of the workforce in the United States by the year 2020.

Members of Generation Y are characterized as smart, tech-savvy, practical, resilient, and as individuals who enjoy the challenge of new opportunities and who value skill development (Lancaster & Stillman, 2002; Zemke, et al., 2000). In line with Marston (2009), the Millennials generation values a highly socialized work culture and embraces cooperative work environments with confidence and enjoyment. Moreover, Generation Y is often associated with the Baby Boomer generation and considered to be optimistic, driven, well-educated, and demanding in the work environment (Lancaster & Stillman, 2002; Skår, Sniehotta, Araújo-Soares, & Molloy, 2008). Additionally, members of Generation Y are digital natives who expect the flow of information to be quick; they embrace organizational change more fluidly than previous generations (Singh, 2013; Valcour, 2013).

**Generation Z:** Generation Z is the next generational cohort to be discussed. There is little research exploration on the characteristics, role, and impact of Generation Z in the workplace due to its recent emergence into the workforce of the United States. Generation Z represents individuals born between the years 1997 and the present (Dimock, 2019). Although a chronological endpoint has not been determined for this age group, this study follows the age range proposed by the pew research center (Dimock, 2019). Per Cabrera (2017), the chronological range of Generation Z coincides with Generation Y, and as a result, there are many shared experiences between these two cohorts. As Cabrera (2017) concludes, some existing literature considers Generation Z as a continuation of Generation Y, a second wave of the previous generation.

Members of the Generation Z cohort are shaped by experiences such as the 9/11 terrorist attack, the Great Recession, worldwide economic decline, and the emergence of

climate change (Cabrera, 2017). Generation Z is said to be tech-savvy like generation Y, inclusive, communal, interconnected in forging a global society, well-learned, and well-cultured (Braz, Frey, Rohr da Cruz, Camargo, & Olea, 2011; Lanier, 2017; Sinclair, 2013). However, since members of Generation Z are just now entering the workforce and little research exists to explore their impact in the work environment, it is difficult to ascertain their readiness for change and how this generation might respond to the organization.

# Generational differences and similarities

Weston (2006) noted that although diversity is commonly identified within the context of race and ethnicity, diversity can also be represented by different generational employees working together. Empirical research suggests there are differences between generational cohorts and generational diversity influences the outlook and results of workers within organizations (Sessa, Kabacoff, Deal, & Brown, 2007). Marshall (2004) asserts that there are differences and similarities among generations; for example, Generation Y is significantly different from its Baby Boomer parents, but both have similar traits to those of Generation X. Among the noted differences is the perception of the work-life balance in all generations (Glass, 2007). There is limited literature on whether or not generational differences affect the readiness for organizational change. As such, the premise of this study is to evaluate the impact of generational cohorts on the readiness for change.

### Generational response to change

Storms (2004) suggested that employees resist change programs because of fear or anxiety of the unknown. Since organizations are increasingly diverse with different

generational perspectives, leaders need to understand the characteristics of the various generations and the most appropriate method for reducing fear and change resistance (Kidwell, 2003). Pihulyk (2003) intimated the causes of change resistance include conflicting personal values, the emotional perspective of the change, and the lack of trust in the change process and leadership. Further, although employees have become more accepting of change initiatives, organizational change programs are still risky and costly because of the lack of employees' readiness for organizational change (Price & Chahal, 2005). The literature review revealed many possible reasons employees resist change, but little is known about the readiness for organizational change in the healthcare environment.

## The impact of tenure and job categories

Much is written on the effects of tenure on organizational commitment and engagement of employees. However, very little is written about the moderating effects of tenure and position on employee readiness for organizational change. Mowday, Porter, and Steers (1982) indicated that individuals behaviorally have a higher initial commitment to an organization, but after that, engage in behaviors that enhance their performance and positional level within the workplace. This assertion indicates that there is a possible effect of an employee's commitment and readiness to engage in change initiatives. Cohen (1991, 1992) states that there is a moderating effect of employee tenure and position groups on organizational commitment and outcomes. Further, Wright and Bonett (2002) suggested that the use of tenure in research is most effective when categorized into career stages and examined with the age difference of employees. However, little is known about the moderating effect of tenure on generational cohorts

when evaluating generational readiness for change. The lack of literature on the impact of tenure and position category and their moderating effect on change readiness provides a premise for this study.

#### Overview of correlational analysis model

The correlational analysis is a statistical method for estimating the relationship among two or more variables and was initially used to study the inheritability of characteristics of peas across generations (Gallo, 2015; Stanton, 2001). Sir Francis Galton conceptualized correlation and the regression model while attempting to predict the characteristics of the progeny of peas from parental characteristics (Azen & Budescu, 2009; Stanton, 2001). Sir Francis Galton develop the idea of correlation and regression when it was observed that the values of characteristics among parental and subsequent generations were closer to the means or averages and was mathematically formalized by Karl Pearson using general techniques of multiple regression and the product-moment correlation coefficient (Azen & Budescu, 2009; Kuiper, 2008; Stanton, 2001).

The concept of correlational analysis enables researchers to identify the correlations among variables by measuring linear relationships (Cooper & Schindler, 2002; Kuiper, 2008; Pedhazur, 1997). The goal of the correlation is to develop a statistical model to describe the relationship between the continuous and dichotomous variables, determine the co-relationship or association of two quantities, identify the direction and strength of association between variables, and determine the extent to which the relationship between variables is linear (Kuiper, 2008; Laerd, 2013; Stanton, 2001). According to Creswell (2005), correlational research is the process and application

of correlation between two or more variables and is useful in understanding and recognizing the best predictors that influence an outcome.

Creswell (2005) further argued that a correlational research design allows researchers to examine the "direction of the correlation of scores, a plot of the distribution of scores to see if they are normally or non-normally distributed, the degree of association between scores, and the strength of the association of the scores" (p. 343). Additionally, correlational studies that identify causal relationships are best used when data is collected using a quantifiable configuration (Gall, Gall, & Borg, 2003). However, causation is not implied through correlational research, and direct associations may exist between variables if a relationship is conclusively determined. For this dissertation research, a correlational analysis was applied to assess the strength and direction of the relationship of position category and tenure with generational cohorts to explain the variable, readiness for organizational change. The correlational design was appropriate to test if a relationship exists between the variables of this study (Cooper & Schindler, 2003).

## Summary

This chapter discussed the literature on organizational change, employee response to change, readiness for change theory, generation theory, characteristics, similarities, and differences of generation cohorts, generational response to change, and offered an overview of the impact of tenure and job categories and multiple regression literature relative to organizational research. This chapter evaluated the resistance, response, and readiness for organizational change and identified the values and characteristics of the Baby Boomer, Generation X, Generation Y, and Generation Z as related to organizational

change. The following, Chapter 3, highlights the research design, sampling methodology, instrumentation, and data collection methodologies used in this dissertation research. Chapter 3 also presents discussions on the theoretical framework, identified variables, sample preparation and handling of the data, and the analysis of the data using selected statistical tests.

### **CHAPTER 3- METHOD**

This chapter focuses on the design of the research and includes discussions about the overall methodology that was applied to the study. In this chapter, the research rationale, research design, population and sampling considerations, instrumentation measures, data collection procedure, and data analysis process is discussed. Additionally, the research questions, sub-questions, hypotheses, and sub-hypotheses are introduced. Finally, ethical considerations, the confidentiality procedure, and the informed consent process is discussed.

Organizations operate within environments which are always influenced by change, and the continuous process of change is costly and has proven to be detrimental for companies (Axley & McMahon, 2006; Burke, 2013; Burnes 2004; Choi & Ruona, 2011; Higgs & Rowland, 2005). However, organizational change promotes innovation and increases the competitive advantage of companies (Axley & McMahon, 2006; Bareil, 2013; Choi & Ruona, 2011). Literature asserted that over 70% of organizational change initiatives fail (Bateh, et al., 2013; Burke, 2010; Burnes 2004; Pellettiere, 2006; Strebel, 2009), that this fail rate could rise to approximately 93% (Decker et al., 2012), and that the high failure rate has been costly for organizations (Higgs & Rowland, 2005; Kotter, 1996; Warrick, 2009). Moreover, several researchers asserted that the increased failure rate of organizational change is due to employees resisting change initiatives because of the lack of readiness, highlighting the need to study the individual readiness for organizational change (Armenakis et al., 1993; Bouckenooghe et al., 2009; Choi & Ruona, 2011).

Armenakis et al. (1993) asserted that the success of change initiatives is highly dependent on the employees' support of and readiness for organizational change. Lewis (2006) argued that the perception of employees impacts the success and resistance of organizational change. A qualitative study conducted by Bourne (2015) suggested that generational cohort perceptions impact the success and resistance of organizational change. Literature asserted that the increasing generational diversity in the workplace resulted in four generations working within the same organizations (Lancaster & Stillman, 2002; Jeffries & Hunte, 2003; White, 2006), and the increasing rate of organizational change initiatives is not decreasing (Kotter, 2002). It is therefore essential to understand the impact of these generational cohorts on organizational change initiatives and determine their potential impact on the success of change initiatives by investigating their readiness for organizational change (Bourne, 2015; Lewis, 2006). The significance of increasing generational differences requires that organization leaders gain an understanding of these differences and distinguish how generational differences impact change resistance in the workplace (Sessa, et al., 2007). Per Zemke, et al., (2000); multigenerational employees with diverse backgrounds demonstrate various characteristics, values, and career outlooks. Since organizations are increasingly diverse with different generational perspectives, leaders are under pressure to identify the appropriate method for reducing change resistance and increasing change readiness (Kidwell, 2003).

This dissertation study investigated the relationship among generational cohort employees and their readiness for organizational change as moderated by tenure and position category. Two theoretical frameworks are emphasized and guided this study: (a)

generation theory (Mannheim, 1952; Strauss & Howe, 1997) and (b) change readiness theory (Armenakis et al., 1993; Bouckenooghe et al., 2009). Generational cohorts, moderated by the tenure and position category of employees, were used to predict the dependent variable, Readiness for Organizational Change, as evidenced by employees' intentional, cognitive, and emotional responses to organizational change. The purpose of this quantitative study was to determine the extent to which tenure and position category moderate generational cohorts' readiness for organizational change in the healthcare industry in the United States, and a) to extend the research on the readiness to change, b) to seek clarification of any influence that generational cohort characteristics have on the response to organizational change by change recipients, and c) determine if position categories and tenure moderate the characteristics of generational cohorts on the readiness for organizational change.

# **Research rationale**

The purpose of this study was to investigate the relationship among generational cohorts, the independent variable, and readiness for organizational change, the dependent variable, among employees with different tenure levels and position categories which serve as the moderating variables. According to Swanson and Holton (2005), quantitative research methods are suitable for use in studying groups and for generalizing to a broader population; the quantitative methods used for this purpose are also suitable for making inferences from smaller groups to larger groups. The applicability of this study to the broader healthcare population will be valuable in contributing to the understanding of the phenomenon of generational cohorts and readiness of organizational change. The scope

of the study was limited to the perspective of healthcare organizations in the United States.

#### **Research design**

Readiness for change theory contends that when employees are better prepared for change, they will be more likely to support it (Armenakis et al., 1993; Bouckenooghe et al., 2009). Employees resist change because of fear, uncertainty, poor communication from management, and lack of resources (Armenakis et al., 1993; Bouckenooghe et al., 2009; Holt et al., 2007; Oreg et al., 2003). The resistance to change has led to a persistently high rate of failure for change initiatives, making them present an undue financial burden on healthcare organizations. The limited availability of literature on generational readiness for change in the healthcare sector necessitates a research design to investigate the influence of generations on change readiness. The study design was a quantitative, cross-sectional exploratory research using correlation analysis to examine and determine statistically significant relationship between the independent and dependent variables (Field, 2009), and between the independent and dependent variables in the presence of moderating variables (Aguinis & Pierce, 2006; Field, 2009; Laerd, 2013).

#### **Research questions, sub-questions, and hypothesis**

The emphasis of this research was to investigate the relationship between generational cohorts and employee readiness for organizational change, and whether or not varying tenure levels and position categories moderated the effects of generational cohorts on employee readiness for organizational change. The omnibus research question and hypotheses of consideration were:

**RQ0:** What is the relationship between generational cohorts (GENCO) and employee readiness for organizational change (ROC) in a healthcare environment when moderated by tenure (TE) and position category (POC)?

**H0:** The generational cohorts will not explain the relationship between the dependent variable, Employee readiness for Change and the four independent variables of generation cohorts when moderated by tenure and position category.

**HA:** There is no statistically significant relationship between generational cohorts and the dependent variable, Employee Readiness for Change, and the four independent variables of generation cohorts when moderated by tenure and position category.

The two research questions (RQ1 and RQ2) and two null and alternative hypotheses (H01,

HA1 and H02, HA2) implied in the omnibus null hypothesis (H0) are as follow:

**RQ1:** What is the relationship between generational cohorts and employee readiness for organizational change?

**H**<sub>01</sub>: There is no statistically significant relationship between generational cohorts and employee readiness for organizational change

**H**<sub>A1</sub>: There is statistically significant relationship between generational cohorts and employee readiness for organizational change.

**RQ2:** What is the relationship among tenure, position category, and generational cohorts predict a relational effect on employee readiness for organizational change?

**H**<sub>02</sub>: There is no statistically significant relationship among the variable of tenure, position category, generational cohorts, and employee readiness for organizational change.

**H**<sub>A2</sub>: There is statistically significant relationship among tenure, position category, generational cohorts, and employee readiness for organizational change.

Sub-question 1 and sub-hypotheses 1. A standard multiple regression analysis was

used to investigate the relationship between the different generational characteristics of

healthcare employees and their readiness for organizational change. The following are

sub-questions (RQs1a - RQs1d) and sub-hypotheses (Hos1a - Hos1d and HAS1a - HAS1d) for

the first research question.

**RQ**<sub>S1a</sub>: What is the relationship between Baby Boomer characteristics and employee readiness for organizational change?

 $H_{0S1a}$ : There is no statistically significant relationship between Baby Boomer characteristics and employee readiness for organizational change.

**H**<sub>AS1a</sub>: There is statistically significant relationship between Baby Boomer characteristics and employee readiness for organizational change.

**RQ**<sub>S1b</sub>: What is the relationship between Generation X characteristics and employee readiness for organizational change?

**H**<sub>0S1b</sub>: There is no statistically significant relationship between Generation X characteristics and employee readiness for organizational change.

**H**<sub>AS1b</sub>: There is statistically significant relationship between Generation X characteristics and employee readiness for organizational change.

**RQ**<sub>S1c</sub>: What is the relationship between Generation Y characteristics and employee readiness for organizational change?

**H**<sub>0S1c</sub>: There is no statistically significant relationship between Generation Y characteristics and employee readiness for organizational change.

**H**<sub>AS1c</sub>: There is statistically significant relationship between Generation Y characteristics and employee readiness for organizational change.

**RQ**<sub>S1d</sub>: What is the relationship between Generation Z characteristics and employee readiness for organizational change?

 $H_{0S1d}$ : There is no statistically significant relationship between Generation Z characteristics and employee readiness for organizational change.

**H**<sub>AS1d</sub>: There is statistically significant relationship between Generation Z characteristics and employee readiness for organizational change.

Sub-question 2 and sub-hypothesis 2. The following are sub-questions (RQs2a –

 $RQ_{s2b}$ ) and sub-hypotheses ( $H_{0s2a} - H_{As2a}$  and  $H_{0s2b} - H_{As2b}$ ) for the second research

question. The interactive effect of the four position categories and tenure levels on

generational cohort characteristics were investigated in relation to the readiness for

organizational change of healthcare system employees. The sub-questions and sub-

hypotheses are as follow:

**RQ**<sub>S2a</sub>: What is the relationship among tenure at the 0-to-2 level, 3-to-5 level, 6-to-8 level, 9+ level, generational cohorts, and employee readiness for organizational change?

**H**<sub>0S2a</sub>: There is no statistically significant relationship among the variables of tenure at the 0-to-2 level, 3-to-5 level, 6-to-8 level, 9+ level, generational cohorts and Employee Readiness for Organizational Change.

**H**<sub>AS2a</sub>: There is statistically significant relationship among the variables of tenure at the 0-to-2 level, 3-to-5 level, 6-to-8 level, 9+ level, generational cohorts and Employee Readiness for Organizational Change.

**RQ**s2b: What is the relationship among the position categories of medical support staff category, administrative staff category, specialty and ancillary service staff category, data management and other position categories, generational cohorts, and employee readiness for organizational change?

**H**<sub>0S2b</sub>: There is no statistically significant relationship among the variables of position category at the medical support staff category, administrative staff category, specialty and ancillary service staff category, data management and other position categories, generational cohorts, and employee readiness for organizational change.

**H**<sub>AS2b</sub>: There is statistically significant relationship among the variables of position category at the medical support staff category, administrative staff category, specialty and ancillary service staff category, data management and other position categories, generational cohorts, and employee readiness for organizational change.

A correlational analysis was used to assess the relationship of the independent

variables on the dependent variable. First, the correlation statistical tool was used to

assess the linear relationships among the four generational cohorts of healthcare

employees and their readiness for organizational change. Second, the correlation analysis

was used to estimate the relationship between the moderated variables of tenure, position

category, and generational cohorts on employee readiness for organizational change. To address the above questions and sub-questions, the dimensions of employee readiness for organizational change was correlated against each generational cohort. The dimensions of employee readiness for organizational change was correlated along with the generational cohorts by each tenure level and position category.

# **Population and sample**

After approval was received from the dissertation committee, the Internal Review Board (IRB), and the School of Business of George Fox University, formal inquiries were made with potential healthcare organizations for the permission for study participants. The selection of data sources for this study was collected from the population and sample group defined below. The population, sampling method, and sample size are described in this section. Additionally, the rationale for selecting the sample size used is explained.

**Population.** A population is a group of individuals who share common traits or characteristics (Vogt, 2009). The population of consideration for this quantitative, cross-sectional exploratory research was the employees working within healthcare systems throughout the Southeastern states of United States of America. These Southeastern states include Alabama, Florida, Georgia, Louisiana, Mississippi, North Carolina, South Carolina, and Tennessee. The population included members of four generational cohorts: (a) Baby Boomers, (b) Generation X, (c) Generation Y, and (d) Generation Z. The population for this research comes from the healthcare industry at healthcare institutions across the Southeastern United States. The sample data was collected across a network of hospitals and medical establishments across the states using electronic surveys distributed

to employees of these institutions. Study participants ranged in age from 18-73. No specific race, gender or cultural background was required to participate in the study. The researchers collaborated with the Qualtrics research department to target the generational groups needed to support this study. The online survey was distributed, and responses collected using the Qualtrics survey platform.

**Sample**. The sample analyzed were between the ages of 18 and 73, and the generational cohorts were moderated by position category and tenure to determine the readiness for organizational change. A random sampling approach and the stratification of intergenerational cohorts by generation was used to collect the data. The sample of interest was working adults ( $\geq$  18 years) who are healthcare employees across the United States of America. The markers that were most consistent with this study were employed adults who had attained high school or higher education, including college or university degrees. The sample size consisted of 200 employed adults who were drawn and stratified by 50 respondents each according to generational cohorts. According to Snedecor and Cochran (1989), a sampling frame is dependent on identifying markers that are most consistent with the purpose of the study being undertaken.

Field (2009) asserts that a common heuristic approach in selecting sample size is for a researcher to sample at least 10–15 participants per variable. Further, Snedecor and Cochran (1989), Field, (2009), and Lehmann and Romano (2005) intimated that large sample size is needed for research, that the sample must be randomly selected, and that the sample size consists of 30 or more participants. Additionally, a priori power analysis for a point biserial model correlation was conducted in G-POWER3 to determine a sufficient sample size for this study using an alpha of 0.05, a power of 0.95, and a

medium effect size ( $f^2 = 0.30$ ) (Faul, F., Erdfelder, E., Buchner, A., & Lang, A.-G., 2013). Based on these assumptions and analysis, the desired total sample size is 134. As such, a total sample size of 200 with 50 participants per the generational cohort variable adequately satisfy the suggestions of these existing literatures and provided adequate sample size in support of this study.

### **Informed consent**

Informed consent is a process which allows researchers to disclose and appropriate information to competent research participants to ensure that participants can make a deliberate choice to accept or refuse participation (Appelbaum, 2007; Crow, Wiles, Heath, and Charles, 2006). Per Crow, et al. (2006), the provision of informed consent enables the researcher to provide adequate information about the research to participants and ensure the participants are provided the opportunity to accept or refuse participation without undue pressures and negative consequences. As such, necessary actions were taken to enable research participants to make informed decisions by providing the full disclosure of the intent and purpose of this dissertation study.

Study participants were provided with a consent form to acknowledge or withdraw their participation. The data for this study were collected through an online survey and participants completed an electronic consent form which communicates (1) their acknowledgement of their voluntary participation in the research effort, (2) their complete understanding of the purpose of the research, and (3) their right to terminate participation in the research at any time, without consequence. The electronic consent form was provided to inform participants of essential research information, provided instructions relating to their rights, provided them with directions to accept or withdraw

participation, and advised participants of the risks and benefits relating to participation (Appelbaum, 2007; Cone & Foster, 2004). The consent form provided included information to ensure confidentiality, the voluntary nature of this dissertation study, and provided the contact information for additional details and questions about the study if needed (Appelbaum, 2007; Cone & Foster, 2004). Copies of informed consent forms were filed and saved electronically in a password protected flash drive (Appendix A).

# Confidentiality

The researcher applied strict principles of confidentiality to protect the privacy and responses of study participants. As stated by Cone and Foster (2004), it is the responsibility of the research to protect the privacy of study participants and maintain confidentiality by protecting the research records and identity of study participants. The confidentiality of participants was maintained through the use of anonymous online surveys and during data collection, storage, and analysis. To further maintain confidentiality and protect the privacy of participants, personally identifiable information was not collected. The collection of demographic information including identifying features such as names, addresses, e-mails or IP addresses, and organization of employment was not collected. To identify the generational group to which participants belong, a date range corresponding to each generational cohort was included in the survey. However, this information cannot be directly linked to or used to identify participants.

## Measures

Producing research results that are valid, trusted, and generalizable to large populations require the use of reliable instruments that can be used to gather data. The

reliability of the instrument must consistently gather data across studies, making it essential for future researchers to replicate. The credibility of the instruments used for research must also provide internal consistency and external validity. Thus, the type of survey questionnaire that was used to collect information on the generational cohorts and the readiness for organizational change is the "R" section of the Organizational Change Questionnaire (OCQ-C, P, R) scale (Bouckenooghe et al., 2009).

The Organizational Change Readiness Questionnaire scale used in this study is divided into a three-questionnaire diagnostic measurement tool consisting of these sections: (C) Climate for Change; (P) Process of Change, and (R) Readiness for Change (Bouckenooghe et al., 2009). For the study, the (R) Readiness for Change portion of the questionnaire is used to understand the resistance and readiness of generational employees. The questionnaire is copyrighted by Bouckenooghe, Devos, and Van den Broeck (2009). The questionnaire was normed for use in for-profit and non-profit organizations to measure the three dimensions of readiness of employees for change. The Organizational Change Readiness Questionnaire original scales manifested significant levels of reliability for the three constructs being investigated: Emotional ( $\alpha = .70$ ), Cognitive ( $\alpha = .69$ ), and Intentional ( $\alpha = .89$ ) (Bouckenooghe et al., 2009). The average reliability coefficient for the original total Readiness for Change scale was 0.76, which indicated adequate, minimally acceptable reliability (Bouckenooghe et al., 2009; Field, 2009; Taber, 2016; Tavakol & Dennick, 2011).

**Reliability.** Cronbach's alpha statistical test is widely used for estimating the reliability of measurement instruments (Vogt, 2007; Taber, 2016; Tavakol & Dennick, 2011). Cronbach's alpha determines if the items on a scale measure the internal
consistency of items on the test. As such, the Cronbach's alpha is a numerical coefficient of reliability that ranges from 0 to 1.0 and, alpha should be .70 to be considered adequate, and above .80 to be considered very reliable (Field, 2009; Taber, 2016; Tavakol & Dennick, 2011). The properties of the Organizational Change Readiness Questionnaire OCQ-R scale revealed an acceptable reliability as follows: Cognitive,  $\alpha = .69$ , Affective/Emotional,  $\alpha = .70$ , and Intentional Readiness for Change;  $\alpha = .89$ ; overall  $\alpha =$ .76, indicating that the scale consistently measures the construct of change readiness within organizations (Bouckenooghe et al., 2009).

**Validity.** The validity of instrument measurements allows for the determinations of the appropriateness and relevance of the research design and measurements. Per Vogt (2007) validity is both internal and external. Internal validity focuses on truthfulness of the research, the accuracy of the conclusions, and the generalizability of the results. The internal validity also ensures that the variables of the research are being measured as intended. External validity ensures that sample selection is representative of the population. As such, the random sampling methods and the large sample size lends to the validity of this research.

### Procedure

The data for this study were randomly collected using Qualtrics.com using the simple random, and criterion sampling methods that were applied are stratified sampling (Snedecor & Cochran, 1989). The services of Qualtrics for data collection was suitable for this study because it provided an efficient and economical means of distributing the survey, receiving responses, allowing for high-security measures, and participant anonymity. Qualtrics is well recognized and widely accepted as a data gathering service

that can be utilized for research in academia and industry. The authorized questionnaire used for data collection was reproduced in Qualtrics. The Qualtrics link to the survey was then disseminated to healthcare employees electronically.

The study participants are working professionals with access to computers and with a basic understanding of computer use. The study participants can reasonably be expected to use computers to complete the survey. Preceding the questionnaire was an electronic consent form to be completed by study participants. The electronic consent form allowed participants to acknowledge their voluntary involvement in this study. The consent form also required the respondents to confirm their understanding of the reason for this research as well as their right to terminate participation without penalty. Personally identifiable information of participants was not collected to protect participant privacy and maintain anonymity. All demographic information that included identifying features such as names, addresses, or e-mails were not collected. However, the gender of the study participants was collected to provide the reader with a picture of the structural makeup of the sample. Additionally, the age group of respondents was collected to identify the generational cohorts belonging to the participants.

### **Data Analysis**

The study was a quantitative, cross-sectional exploratory research using standard multiple regression analysis and mediated multiple regression analyses to test the null hypotheses using IBM's SPSS analysis tool. The correlation analysis tool was used to investigate the relationship between generational cohorts and readiness for organizational change. The correlation test was also used to investigate the hypothesis that within tenure level and position categories there is no relationship between generation cohorts and

readiness for organizational change. Additionally, a correlation matrix was also generated to assess the relationship between readiness for change and Baby Boomers, Generation X, Generation Y, and Generation Z. To measure the internal consistency of the construct of this study, a descriptive statistic, frequency, and an exploratory data analysis command were run to summarize data information. A reliability check was also performed to test the function of the internal consistency of the study constructs. Data information and statistics were summarized to observe the number of cases, the mean, standard deviation, range, skewness, and kurtosis of the dataset. The frequency command provided vital visual data about the demographic makeup, skewness, and kurtosis of the dataset. The exploratory data analysis information used provided information about missing data and outliers (Field, 2009).

## **Ethical considerations**

Ethical concerns involving this study were addressed throughout the research process because of the involvement of human subjects (Creswell, 2009). This study was established on the ethical principles of beneficence, respect for persons, and justice. These ethical considerations guided the moral actions of the researcher. However, no known risks or ethical issues were identified that impacted the target population of the study, the sample, or the online questionnaire which was voluntarily accessed. Participants were treated with respect as self-directed and voluntary participants who were provided with informed consent before participation in the survey. Considering the beneficence, the welfare of the participant was of most importance. As such, the participating human subjects were not harmed emotionally or physically, and all possible benefits were maximized individually. Furthermore, no known risks were identified. The

research maintained the principle of justice by providing an equal level of engagement and anonymity throughout the study. Qualtrics.com was used for data collection for this study. This study was reviewed by the researcher's dissertation committee and approved by the Institutional Review Board (IRB) of George Fox University before beginning data collection. In accordance with the ethical principles established by the Nuremberg Code, the Helsinki Declaration, and the Belmont Report, the principles of respect for persons, beneficence, and justice was upheld.

# **Summary**

This chapter described the research design, rationale, and methodologies that were applied to this dissertation study. A quantitative, non-experimental, cross-sectional design was utilized to support this research. Further, random sampling techniques were utilized to identify the criterion-based sample from working adults within a healthcare environment in the United States of America. The descriptions of the instrumentation and statistical measures used to test the hypotheses were explained along with the informed consent, confidentiality, and ethical considerations. The research questions, subquestions, null hypotheses, and sub-hypotheses were presented. In the following chapter (Chapter 4), the results of this dissertation research are presented. The analytical procedures, data analysis and interpretation, and discussions of the findings of this study are explored.

### **CHAPTER 4- RESULTS**

This chapter presents the results generated from this cross-sectional quantitative research. The overall intent of this research was (a) to investigate the relationship of Generational Cohort (Baby Boomers, Generation X, Generation Y, and Generation Z) characteristics with Employee Readiness for Organizational Change, (b) to determine if the Tenure Levels and Position Categories of generational employees contributed to the Readiness for Organizational Change, and if so, (c) to assess the strength and direction of the relationship between the variables. In this chapter, the research methodology and the results of the study are presented. In this cross-sectional correlational design, the four generational cohorts (GENCO) were tested to evaluate whether or not a significant relationship exists between the employee readiness for organizational change (ROC) and the moderating relationship of generational employee tenure (TE) and position category (POC).

The three dimensions of organizational change readiness (cognition, emotion, and intention) comprised the scale that indicated and measured employee readiness for organizational change, and the three sub-scales gathered data on these dimensions of ROC. A correlational analysis was used to answer the research questions and subquestions and test the hypotheses and sub-hypotheses that were generated from the research questions. These research methodology procedures investigated the relationships between generational cohorts and employee change readiness and evaluated the possible impact of position category and tenure levels on generational employees' change readiness. Additionally, the research procedures sought to explain the direction of the

relationship and how much variance in readiness for organizational change might be explained by generational characteristics.

In the following sections, the details of the data analysis and results are discussed with the research questions and hypotheses. However, data collection, a description of the population and sample, description of the participants, the research design and methodology, instrument measures, instrument reliability and viability are first presented.

### **Data Collection**

The data for this study were randomly collected using one survey instrument. Demographic questions were specified and included in the instrument to collect demographic data such as gender, education, geographic region, employment tenure levels, and generational age range of study participants. To maintain anonymity, no personally identifying information such as name, address, IP address or phone number was collected. A validated and reliable questionnaire, the OCQ-R section of the three-part OCQ-C, P, R questionnaire, was used to collect data on the employee readiness for organizational change. The OCQ-R section of the Organizational Change Questionnaire contains nine questions, and three of the questions were reverse scaled. The OCQ-R questions measured the three dimensions of employee readiness for change: cognitive readiness, emotional readiness, and intentional readiness. Qualtrics.com survey platform was used to collect survey responses, and the latest version of IBM's SPSS® Statistics software was used to analyze the data. The data was collected and transferred into the SPSS software for data analysis.

### **Description of the Population and Sample**

The sample was collected using an electronic survey platform. For this research, the services of Qualitrics.com were employed to distribute the surveys to a randomly selected criterion-based sample from a population of voluntarily registered adult healthcare employees throughout the Southeastern United States of America (USA). Adult participants were recruited from the southern states of Alabama, Florida, Georgia, Louisiana, Mississippi, North Carolina, South Carolina, and Tennessee. Research participants were given the opportunity to make voluntary informed decisions regarding their participation. Study participants were required to read and electronically acknowledge that they understood the description of the study as outlined as provided on the electronic informed consent form. Study participants were informed that they could discontinue participation in the survey without consequences. The informed consent form explained how participants are protected from harm by outlining the risks and benefits of participating in this research. Additionally, the informed consent form also provided emphasized how the information that they provided would be protected. Participants who provided consent were directed to the survey for completion. Those participants who declined consent were redirected to an exit page and were not given the opportunity to complete the survey. The response of participants who did not complete the survey in its entirety was not included in this study.

### Sample Frame and Sample Size

In quantitative studies, the sample size is guided by heuristics, such as 10, 15, or 30 participants per variable (Field, 2009; Aguinis, 2010). A priori G-Power analysis for a point biserial model correlation conducted in G-POWER3 with an alpha of 0.05, a power

of 0.95, and a medium effect size ( $f^2 = 0.30$ ) was used to determine sample size (Faul, F., Erdfelder, E., Buchner, A., & Lang, A.G., 2013), and the desired total sample size for this study was 134. A sample frame of adults employed in the healthcare industry in the southeastern United States of America was targeted for this research, and a sample of 210 actively employed individuals who were stratified by generational cohorts was randomly drawn. There was 100% (n = 210) response rate without missing responses. From this sample size, 53 participants were assigned to the Baby Boomer generation, 53 participants were assigned to Generation X, 53 participants were assigned to Generation Y, and 51 participants were assigned to Generation Z.

The sample size consisted of participants with educational levels which range from associate degrees to doctoral degrees. Unemployed adults who satisfied the generational cohort criterion were excluded from the study, as well as participants under the age of 18 (<18years old) who may have otherwise satisfied this criterion. Additionally, participants who failed to complete the survey in its entirety and those who were not employed in the healthcare setting in the Southeastern states of the United States were excluded from the research.

# **Effect Size**

Acceptable sample size in quantitative research is often guided by heuristics such as 10, 15, or 30 participants per variable (Field, 2009; Nunnally, 1977). However, Aguinis and Gottfredson (2010) and Nunnally (1977) recommended that larger sample sizes ranging from 200 to 400 for quantitative analysis. Aguinis and Gottfredson. (2010) further suggests that using a larger sample size in quantitative analysis allowed for the accurate detection of effect sizes, allowing researchers to avoid a Type 1 error and accept

a null hypothesis when it should be rejected. J. Cohen (1992, 1988) suggested that the correlation coefficients can be used to determine the effect size as follows, R=0.10 (Small effect): In this case, the effect explains 1% of the total variance; R=0.30 (Medium effect): In this case, the effect explains 9% of the total variance; R=0.50 (Large effect): In this case, the effect explains 25% of the total variance. For this current study, the correlations coefficient was used to determine the effect size.

## **Description of Participants**

For this dissertation research, a heterogeneous, randomly selected sample was used. Participants were randomly drawn from eight Southeastern states including Alabama, Florida, Georgia, Louisiana, Mississippi, North Carolina, South Carolina, and Tennessee, to ensure regional diversity. Two hundred and ten (n = 210) participants were stratified by generational cohorts and were assigned to three groups of 53 and one group of 51 based on these criteria. The composition of the generational cohort (Table 1) was as follows: 1946 to 1964 (Baby Boomer Generation)(n = 53) who made up 25.2% of the sample; 1965 to 1980 (Generation X (n = 53, 25.2%), 1981 to 1996 (Generation Y (n =53, 25.2%), and 1997 to later (Generation Z (n = 51, 24.3%). Additionally, the gender makeup of the population was predominated by women, who make up 83.8% (n = 176) of study participants, and this composition was not representative of the general population in the United States or the Southeastern states. According to its most recent supplemental report (2017), the US Census Bureau recorded the gender makeup of the US as 49.2% male and 50.8% female. The female participation in this study far exceed those of males by almost six times in the sample.

| Gender                          | Frequency | Percent | Valid<br>Borecont | Cumulative            |
|---------------------------------|-----------|---------|-------------------|-----------------------|
|                                 |           |         | rercent           | rercent               |
| Male                            | 34        | 16.2    | 16.2              | 16.2                  |
| Female                          | 176       | 83.8    | 83.8              | 100.0                 |
| Total                           | 210       | 100.0   | 100.0             |                       |
| Generational<br>Cohorts (GENCO) | Frequency | Percent | Valid<br>Percent  | Cumulative<br>Percent |
| Baby Boomer                     | 53        | 25.2    | 25.2              | 25.2                  |
| Generation X                    | 53        | 25.2    | 25.2              | 50.5                  |
| Generation Y                    | 53        | 25.2    | 25.2              | 75.7                  |
| Generation Z                    | 51        | 24.3    | 24.3              | 100.0                 |
| Total                           | 210       | 100.0   | 100.0             |                       |

Table 1: Population by Gender and Generational Cohorts

Of participants' tenure level (table 2), 32.9% (n = 69) the participants in this study worked at their healthcare organization with a tenure level of 0 - 2 years while 21.4% (n = 45) of the participants have been employed for with their organization for 3 - 5 years. In the 6 - 8 years and 9+ years of tenure level groups, the length of employment for participants is 15.2% and 30.5%, respectively. Additionally, from table 2, observe that 54.8% (n = 115) of the participants in this study were medical support staff, which predominated the population.

Table 2: Population by tenure level and Position Categories

| Tenure Level (TE) | Frequency | Percent | Valid<br>Percent | Cumulative<br>Percent |
|-------------------|-----------|---------|------------------|-----------------------|
| 0 - 2 years       | 69        | 32.9    | 32.9             | 32.9                  |
| 3 - 5 years       | 45        | 21.4    | 21.4             | 54.3                  |

|                                |           |         |         | _          |
|--------------------------------|-----------|---------|---------|------------|
| 6 - 8 years                    | 32        | 15.2    | 15.2    | 69.5       |
| 9 + years                      | 64        | 30.5    | 30.5    | 100.0      |
|                                | 210       | 100.0   | 100.0   |            |
| Total                          | 210       | 100.0   | 100.0   |            |
| <b>Position Category (POC)</b> | Frequency | Percent | Valid   | Cumulative |
|                                |           |         | Percent | Percent    |
| Medical Support                |           |         |         |            |
| Staff (Doctors,                | 115       | 54.8    | 54.8    | 54.8       |
| Nurses, and                    |           |         |         |            |
| Technicians staff)             |           |         |         |            |
| Administrative Staff           |           |         |         |            |
| (Executives,                   |           |         |         |            |
| Managers,                      |           |         |         |            |
| Supervisors,                   | 50        | 23.8    | 23.8    | 78.6       |
| Finance, HR, and               |           |         |         |            |
| organization                   |           |         |         |            |
| operations starr)              | 1         |         |         |            |
| Specialty &                    |           |         |         |            |
| Ancillary Services             | 17        | 0.1     | 0.1     | 967        |
| Staff (Laboratory,             | 1/        | 8.1     | 8.1     | 80.7       |
| other lay staff)               |           |         |         |            |
| Data Managamant                |           |         |         |            |
| Se Other Staff (IT             |           |         |         |            |
| HIM and Other                  |           |         |         |            |
| organizational staff           | 28        | 13.3    | 13.3    | 100.0      |
| not mentioned                  |           |         |         |            |
| above)                         |           |         |         |            |
| ́                              | 010       | 100.0   | 100.0   |            |
|                                | 210       | 100.0   | 100.0   |            |
| Education                      | Frequency | Percent | Valid   | Cumulative |
|                                |           |         | Percent | Percent    |
| Associates level               | 105       | 50.0    | 50.0    | 50.0       |
| Bachelors level                | 68        | 32.4    | 32.4    | 82.4       |
| Masters level                  | 24        | 11.4    | 11.4    | 93.8       |
| Doctoral level                 | 13        | 6.2     | 6.2     | 100.0      |

| Total | 210 | 100.0 | 100.0 |  |
|-------|-----|-------|-------|--|
|-------|-----|-------|-------|--|

From table 2, it is observed that 50.0% (n = 105) of the participants in this study were educated up to associates level, 32.4% (n = 68) were educated up to bachelors level, 11.4% (n = 24) were educated up to masters level, and 6.2% (n = 13) were educated up to doctoral level. As of 2017, in the general US population of over 247 million adults 18 years and older, over 188 million adults had obtained an education that ranged from a high school diploma to a graduate or professional degree. The educational level result from table 2 is not representative of the general population in the United States or the Southeastern states.

### **Research Design and Methodology**

This research applied a quantitative methodology, correlation analysis, to test the relationship, and the strength and direction of the relationship between several variables (Generational Cohorts (GENCO) – Baby Boomer generation, Generation X, Generation Y, Generation Z) and the dependent variable, Readiness for Organizational Change (ROC). This study also sought to examine the relationship in ROC that may have emerged as a result of the tenure level and position category of the different generations. How much the independent variables correlate to determine the strength and direction of the relationship in the dependent variable was what was being examined. Correlational analyses were applied to test the hypotheses and answer the research questions that were generated from the following omnibus research question:

What is the relationship between generational cohorts (GENCO) and employee readiness for organizational change (ROC) in a healthcare environment when moderated by tenure (TE) and position category (POC)?

Random sampling techniques and Likert scale survey instruments were used to collect data to ensure that the researcher did not interact with survey respondents, allowing objectivity. Further, the makeup of the omnibus research question supported the research design and correlational analysis methodology. This is because correlations analysis is used to develop a statistical model to describe the relationship between the variables, determine the co-relationship or association of two quantities, identify the direction and strength of association between variables, and determines the extent to which the relationship between variables is linear (Kuiper, 2008; Laerd, 2013; Stanton, 2001). As such, the correlational design was used to test the existence of a relationship between the variables of this research.

# Measures

The Organizational Change Questionnaire survey instruments were used to electronically collect data concerning generational employee readiness for organizational change from a randomly chosen sample of employed adults between the age range of 18-73 years across the Southeastern states of the USA. The Organizational Change Questionnaire is part of a three-part questionnaire (OCQ-C, P, R) that measures cognitive, emotional, and intentional readiness for change of employees (Bouckenooghe, et al.,2009). As discussed in Chapter 3, the OCQ-C, P, R construct was normed for organizational research. Dr. Dave Bouckenooghe, one of the creators of the OCQ-C, P, R, granted the researcher permission to use questionnaire for this research. The questionnaire was developed to measure an organization's change climate, change process, and change readiness at the organizational and individual levels. However, the three components of the instrument could be used independently of each other to measure

the three independent constructs within an organization. The OCQ-R construct of the scale measures was used for this study because it is designed to measure readiness for organizational change and because the OCQ-R section was developed to measure the individual employee readiness for organizational change, the focus of this study. The OCQ-R construct is composed of three sub-scales that measure employees' cognitive, emotional, and intentional readiness for organizational change. The responses were collected using 5-point Likert scales with the following ranges: 1 = strongly disagree, 2 = disagree, 3 = neither agree nor disagree, 4 = agree, and 5 = strongly agree. However, one sub-scale, the cognitive readiness for change, was reverse scaled as follow: 5 = strongly disagree, 4 = disagree, 3 = neither agree nor disagree, 2 = agree, and 1 = strongly agree. The original OCQ-R scale exhibited acceptable reliability properties overall ( $\alpha$  = .76), and OCQ-R sub-scales individually exhibited acceptable reliability properties (Cognitive Readiness,  $\alpha$  = .69, Emotional Readiness,  $\alpha$  = .70, and Intentional Readiness for Change;  $\alpha$  = .89).

### **Instrument Reliability and Viability**

The reliability and validity of the Organizational Change Questionnaire (OCQ-C, P, R) survey instruments have been established in existing literature and explained in detail in Chapter 3. Per Field (2009) and Vogt (2007), the reliability of an instrument highlights its consistency in measuring what it is supposed to measure and reveals whether or not a research design can be replicated by future researchers. The reliability of an instrument contributes to the validity of the results. Here, when an instrument fails to measure that which it is intended to measure, results generated will not be considered valid. As such, the documented reliability of the original instrument, as discussed above, contributed to

its validity. The reliability in statistics describes the inter-item consistency. For this study, the inter-item consistency in statistics was measured in terms of Cronbach's alpha, which ranged from 0.76 to 0.90 and is deemed to be good. Hence, all the items were considered for further analysis.

| Variables                        | Cronbach's alpha (α) |
|----------------------------------|----------------------|
| Overall                          | 0.84                 |
| Intentional readiness for change | 0.89                 |
| Cognitive readiness for change   | 0.76                 |
| Emotional readiness for change   | 0.90                 |

Table 3: Reliability analysis of OCQ-R scale instrument

The OCQ-R scale used for this study exhibited consistent properties when compared to the original OCQ-R scale (Table 3). The properties of the OCQ-R scale are as follows: The Cognitive Readiness for Change was  $\alpha = .76$ ; the Emotional Readiness for change was  $\alpha = .90$ , and the Intentional Readiness for Change was  $\alpha = .89$ . The overall reliability average was  $\alpha = .84$  These reliability results indicate that the instrument used for this research was valid.

In this study, the OCQ-R scale exhibited better properties than the original instrument overall, which indicated that the items were consistent in measuring what they were supposed to measure. This result supports the validity of the instrument. Furthermore, differences in scale reliability from the original report and from one study to study can be affected by sample size and the composition of the dataset. The evidence of the instruments' ability to measure what it is supposed to measure is determined by examining the statistics in the correlation matrices of the variables. Additionally, a low statistical correlation indicates that no construct in the variables is correlated with each other. Therefore, the constructs in the variables are not redundant and thus ensures discriminant validity.

#### **Details of Data Analysis and Results**

The purpose of this quantitative cross-sectional exploratory study seeks a) clarification of any influence that generational cohort characteristics have on the response to organizational change by change recipients, and b) to determine if position categories and tenure impact the relationship of generational cohorts on the readiness for organizational change among actively employed adults ( $\geq 18$  years old) in healthcare organizations in the Southeastern states of the United States. In other words, the two goals of this study were (a) to determine if the independent variables, the generational cohorts' (GENCO) characteristics influence the dependent variable Readiness for Organizational Change (ROC) by examining how the relationship and direction of the relationship in the dependent variable can be explained by the independent variables, and (b) to determine whether or not the Tenure Level (TE) and Position Category (POC) of GENCO influence employee Readiness for Organizational Change. It was hypothesized that generational cohort characteristics would not have a statistically significant effect on employee readiness for organizational change (H<sub>0</sub>1), and the influence of TE and POC on generational cohorts would not predict a statistically significant relationship in employee readiness for organizational change (H02). To test these study objectives, a correctional analysis was used to the proposed hypotheses ( $H_01$  and  $H_02$ ). Correlational tests are used to investigate the relationship and the direction of the relationships between variables.

# **Means and Standard Deviations**

The mean value of intentional readiness was higher for respondents in iGEN cohort (SD = .80), the mean value of cognitive readiness was higher among participants in GENY cohort (SD = .77), and the mean value of emotional readiness was higher in respondents in the GENX cohort (SD = .76), and the mean value for BABO was statistically consistent but lower for intentional readiness (SD = .78) and emotional readiness (SD = .95) dimensions (table 4).

*Table 4: Mean and Standard Deviation of variables and the dimensions of Change Readiness* 

| Generational Cohorts |                | IRC    | CRC    | ERC    |
|----------------------|----------------|--------|--------|--------|
|                      | Mean           | 3.8302 | 3.6101 | 3.6226 |
| BABO                 | Ν              | 53     | 53     | 53     |
|                      | Std. Deviation | .78344 | .77276 | .95192 |
|                      | Mean           | 4.0189 | 3.5975 | 4.0943 |
| GENX                 | Ν              | 53     | 53     | 53     |
|                      | Std. Deviation | .81496 | .96600 | .76062 |
|                      | Mean           | 4.0314 | 3.7044 | 3.9371 |
| GENY                 | Ν              | 53     | 53     | 53     |
|                      | Std. Deviation | .89458 | .76978 | .79275 |
|                      | Mean           | 4.1307 | 3.5556 | 3.9477 |
| iGEN                 | Ν              | 51     | 51     | 51     |
|                      | Std. Deviation | .80022 | .81012 | .73143 |
|                      | Mean           | 4.0016 | 3.6175 | 3.9000 |
| Total                | Ν              | 210    | 210    | 210    |
|                      | Std. Deviation | .82588 | .82955 | .82688 |

**Mean and Standard Deviation Report** 

# Hypothesis one (H<sub>0</sub>)

The first hypothesis was evaluated to determine how GENCO explains the relationship between ROC when correlated with variables TE and POC. In order to test

the hypothesis, a correlation analysis was applied by using SPSS. There was a significantly negative correlation between the tenure level in healthcare organizations and generational characteristics. However, none of the other correlations demonstrated a statistically significant relationship (Table 5).

|       |                        | GENCO | ROC  | TE    | POC  |
|-------|------------------------|-------|------|-------|------|
| GENCO | Pearson<br>Correlation | 1     | .100 | 700** | .030 |
|       | Sig. (2-tailed)        |       | .149 | .000  | .667 |
|       | Ν                      | 210   | 210  | 210   | 210  |
| ROC   | Pearson<br>Correlation | .100  | 1    | 098   | .109 |
|       | Sig. (2-tailed)        | .149  |      | .158  | .114 |
|       | Ν                      | 210   | 210  | 210   | 210  |
| TE    | Pearson<br>Correlation | 700** | 098  | 1     | 047  |
| IE    | Sig. (2-tailed)        | .000  | .158 |       | .500 |
|       | Ν                      | 210   | 210  | 210   | 210  |
| РОС   | Pearson<br>Correlation | .030  | .109 | 047   | 1    |
|       | Sig. (2-tailed)        | .667  | .114 | .500  |      |
|       | Ν                      | 210   | 210  | 210   | 210  |

*Table 5: Correlational analysis of Tenure Level, Position Category, Generations Cohort, and Change Readiness* 

\*\*. Correlation is significant at the 0.01 level (2-tailed).

As such, the null hypothesis, which asserted that generational cohorts would not explain the relationship between the dependent variable, readiness for organization and the depended variable, generation cohorts, was supported. For this analysis, the effect size was medium with ROC and small with other variables.

Hypothesis two (H<sub>0</sub>1)

Hypothesis 2 evaluated the extent to which generation cohorts predicted a relational effect on employee readiness.

|       |                     | GENCO | ROC  |
|-------|---------------------|-------|------|
| GENCO | Pearson Correlation | 1     | .100 |
|       | Sig. (2-tailed)     |       | .149 |
|       | Ν                   | 210   | 210  |
| ROC   | Pearson Correlation | .100  | 1    |
|       | Sig. (2-tailed)     | .149  |      |
|       | Ν                   | 210   | 210  |

Table 6: Generational cohorts correlated with employee change readiness

From table 6 above, it was observed that the correlation coefficient between the change readiness and generation was 0.100 and its corresponding p-value was 0.149>0.05. Since the p-value is more than 0.05, we can observe that there is no significant association between readiness for change and generation cohorts; as such, the null hypothesis is supported. For this analysis, the effect size was medium for ROC.

# **Hypothesis three (H**<sub>0</sub>**2**)

Evaluation of hypothesis 3, which examined the extent to which tenure and position category interact with generational cohorts to have a relations effect on readiness for organizational change. There was significantly negative correlation between the tenure of work in health care organizations and generations. None of the other correlations were statistically significant. In order to examine the correlation between generation and position categories, Pearson correlation test was applied by using SPSS (Table 7).

|       |                     | GENCO | POC  |
|-------|---------------------|-------|------|
| GENCO | Pearson Correlation | 1     | .030 |
|       | Sig. (2-tailed)     |       | .667 |
|       | Ν                   | 210   | 210  |
| РОС   | Pearson Correlation | .030  | 1    |
|       | Sig. (2-tailed)     | .667  |      |
|       | Ν                   | 210   | 210  |

Table 7: Generation cohorts correlated with position category

The correlation coefficient between the generational cohort and position categories was 0.03, and its corresponding p-value was 0.667>0.05. Since the p-value was more than 0.05, there was no significant correlation between the generation and position categories, yielding a small effect size. In order to examine the correlation between generation and tenure levels, Pearson correlation test was applied by using SPSS (Table 8).

Table 8: Correlation analysis of generation cohorts and tenure level

|       |                     | GENCO | TE    |
|-------|---------------------|-------|-------|
|       | Pearson Correlation | 1     | 700** |
| GENCO | Sig. (2-tailed)     |       | .000  |
|       | Ν                   | 210   | 210   |
|       | Pearson Correlation | 700** | 1     |
| TE    | Sig. (2-tailed)     | .000  |       |
|       | Ν                   | 210   | 210   |

\*\*. Correlation is significant at the 0.01 level (2-tailed).

The correlation coefficient between the generational cohort and tenure was -0.700, and its corresponding p-value was 0.000<0.05. Since the p-value was less than 0.05, there was a significant correlation between the generation and tenure with a small effect size. In

order to examine the correlation between generation and intentional, cognitive, and emotional, the Pearson correlation test was applied by using SPSS (Table 9).

|       |                        | GENCO | IRC    | CRC    | ERC    |
|-------|------------------------|-------|--------|--------|--------|
|       | Pearson<br>Correlation | 1     | .124   | 007    | .112   |
| GENCO | Sig. (2-tailed)        |       | .074   | .922   | .107   |
|       | Ν                      | 210   | 210    | 210    | 210    |
| IRC   | Pearson<br>Correlation | .124  | 1      | .248** | .469** |
|       | Sig. (2-tailed)        | .074  |        | .000   | .000   |
|       | Ν                      | 210   | 210    | 210    | 210    |
| CDC   | Pearson<br>Correlation | 007   | .248** | 1      | .386** |
| CKC   | Sig. (2-tailed)        | .922  | .000   |        | .000   |
|       | Ν                      | 210   | 210    | 210    | 210    |
|       | Pearson<br>Correlation | .112  | .469** | .386** | 1      |
| ERC   | Sig. (2-tailed)        | .107  | .000   | .000   |        |
|       | Ν                      | 210   | 210    | 210    | 210    |

*Table 9: Correlation analysis of generational cohorts, internal readiness for change, cognitive readiness for change, and emotional readiness for change* 

\*\*. Correlation is significant at the 0.01 level (2-tailed).

The correlation coefficient between the generational cohorts and intentional readiness was 0.124, and its corresponding p-value was 0.074>0.05. Since the p-value was more than 0.05, there was no significant correlation between the generational cohorts and intentional readiness. The correlation coefficient between the generational cohorts and cognitive readiness was -0.007, and its corresponding p-value was 0.922>0.05. Since the p-value was more than 0.05, there was no significant correlation between generations and cognitive readiness. The correlation coefficient between the generational cohorts and cognitive readiness. The correlation coefficient between the generational cohorts and cognitive readiness. The correlation coefficient between the generational cohorts and cognitive readiness. The correlation coefficient between the generational cohorts and cognitive readiness. The correlation coefficient between the generational cohorts and cognitive readiness. The correlation coefficient between the generational cohorts and cognitive readiness. The correlation coefficient between the generational cohorts and emotional readiness was 0.112, and its corresponding p-value was 0.107>0.05. Since the

p-value was more than 0.05, there was no significant correlation between generational cohorts and emotional readiness. For the correlation between the GENCO and CRC and between the GENCO and ERC, the effect size was small, and; for the correlation between GENCO and IRC, the effect size was medium.

# **Regression Analysis and ANOVA**

A statistically significant correlational relationship was not observed between generational cohort characteristics and the three dimensions of organizational change readiness. As such, regression analysis and general linear model analysis was applied using SPSS to test the between-subject effects and to determine if there is a significant relationship between generational characteristics and change readiness. The results of the regression analysis indicate that the beta coefficient between the readiness and generation was 0.100 and its corresponding p-value was 0.149>0.05 (Table 10).

Table 10:Regression analysis of change readiness and generational cohorts

# **Coefficients**<sup>a</sup>

| Model      | Unstandardized<br>Coefficients |            | Standardized | t      | Sig. |
|------------|--------------------------------|------------|--------------|--------|------|
|            | Coefficients                   |            | Coefficients |        |      |
|            | В                              | Std. Error | Beta         |        |      |
| (Constant) | 3.700                          | .106       |              | 34.930 | .000 |
| GENCO      | .056                           | .039       | .100         | 1.450  | .149 |

a. Dependent Variable: Change Readiness

Table 11: General linear model analysis of GENCO, TE, POC, and ROC

# **Tests of Between-Subjects Effects**

Dependent Variable: Change Readiness

| of Squares | Source | Type III Sum<br>of Squares | df | Mean Square | F | Sig. |
|------------|--------|----------------------------|----|-------------|---|------|
|------------|--------|----------------------------|----|-------------|---|------|

| Corrected Model | 4.825ª   | 14  | .345     | .863     | .600 |
|-----------------|----------|-----|----------|----------|------|
| Intercept       | 1169.551 | 1   | 1169.551 | 2927.133 | .000 |
| POC             | .775     | 3   | .258     | .646     | .586 |
| TE              | .838     | 3   | .279     | .699     | .554 |
| POC * TE        | 2.602    | 8   | .325     | .814     | .591 |
| Error           | 77.913   | 195 | .400     |          |      |
| Total           | 3178.802 | 210 |          |          |      |
| Corrected Total | 82.738   | 209 |          |          |      |
|                 |          |     |          |          |      |

a. R Squared = .058 (Adjusted R Squared = -.009)

The p-value is more than 0.05, and there is no significant association between readiness to change and generations. Additionally, the F value for the interaction between organizational change readiness, tenure, position category, and generational cohorts was 0.814, and its corresponding p-value was 0.591>0.05. Since the p-value is more than 0.05, there is no significant association between change readiness and generational cohorts.

*Table 12:Analysis of Variance between the dimensions of readiness and generational cohorts* 

|   |                   | Sum of<br>Squares | df  | Mean<br>Square | F     | Sig. |
|---|-------------------|-------------------|-----|----------------|-------|------|
| Intentional readiness<br>for change questions | Between<br>Groups | 2.471             | 3   | .824           | 1.211 | .307 |
|   | Within<br>Groups  | 140.085           | 206 | .680           |       |      |
|   | Total             | 142.555           | 209 |                |       |      |

ANOVA

| Cognitive readiness                      | Between<br>Groups | .620    | 3   | .207  | .297  | .827 |
|--|-------------------|---------|-----|-------|-------|------|
| for change questions                     | Within<br>Groups  | 143.205 | 206 | .695  |       |      |
|  | Total             | 143.825 | 209 |       |       |      |
| Emotional readiness for change questions | Between<br>Groups | 6.268   | 3   | 2.089 | 3.150 | .026 |
|  | Within<br>Groups  | 136.632 | 206 | .663  |       |      |
|  | Total             | 142.900 | 209 |       |       |      |

The analysis of variance between the different dimensions of change readiness and generational cohort characteristics was statistically not significant for intentional readiness and cognitive readiness and significant for emotional readiness.

### Summary

This chapter (Chapter 4) presented the results of a study that applied correlation procedures to assess the relationship among Generational Cohorts, Tenure, Position Categories, and Readiness for Organizational Change. The results of the correlational analysis indicated that a positive statistically significant relationship between Generational Cohorts characteristics and Readiness for Organizational Change does not exist; therefore, the null hypotheses were supported. Moreover, the results of the correlational analysis indicated that no significant relationship existed when generational cohorts, tenure, and position category was correlated with the Readiness for Organizational Change dimensions. However, the tenure level and position category showed a negatively significant relationship. The descriptive statistics presented highlights the composition of the sample used in this study. The makeup of the sample (n = 210) was criterion-based, and the sample was randomly drawn. Gender, educational level, and generational cohorts were included in the makeup of the sample. However, when evaluating the gender makeup, women predominated men as survey respondents. Further, the reliability statistics differed from those of the original scales with stronger properties than originally reported, supporting the validity of the instrument. Chapter 5, presents the results in the preceding sections, including implications of the study to organizations, individuals, recommendations for future studies, and concluding remarks.

## **CHAPTER 5- DISCUSSION**

This chapter explains the results of the study and draws conclusions from the literature review, research method, and data analysis. The purpose of this correlational cross-sectional study was to explore relationship among generational cohorts (Baby Boomers, Generation X, Generation Y, and Generation Z), tenure, position category, and organizational change readiness. Chapter Five highlights the significance of the study, restatement of the description of the sample, discussions of the findings, implication of the study, limitations of the study, recommendations for future research, and concluding remarks.

Studies have examined the relationship between the readiness for change, organizational workforce, and the success of change initiatives. The literature intimate that organizational change initiatives fail because of the lack of readiness for change which results in employee change resistance (Bouckenooghe et al., 2009; Burnes, 2004). A review of the organizational change literature revealed that more than 70% of all organizational change initiatives fail (Bateh, et. al., 2013; Burke, 2010; Kotter, 1996; Warrick, 2009). Additionally, the organizational change literature further suggests that resistance to change is a behavioral response which is ambivalently manifested in support of or resistance to the change initiatives (Armenakis et al., 1993; Lamm & Gordon, 2010; Oreg & Sverdlik, 2011; Van Egeren, 2009). While studies continue to investigate the correctness and accuracy of the failure rate of change readiness, many organizations continue to hold interest in understanding the reason behind the failure rate because of the high cost associated with the failure rate (Bouckenooghe et al., 2009). The literature also asserted that the responses (resistance to or support for change initiatives) of employees

relating to readiness for organizational change is shaped by the beliefs, attitudes, and intentions which are characterized by the three dimensions of readiness: emotion, cognition, and intention (Piderit, 2000).

Consequently, the literature shows that there is increasing generational diversity of the modern workforce, and this phenomenon continues to challenge organizational (healthcare) leaders as they navigate change initiatives (Lancaster & Stillman, 2002; Sharma, et al., 2018). The increasing generational diversity in the workplace has led to four generations with different characteristics and life experiences working within the same organizations (Lancaster & Stillman, 2002; Jeffries & Hunte, 2003; White, 2006). The multigenerational employees come from diverse backgrounds and demonstrate various characteristics, preferred communication styles, values, and career outlooks, which are referred to as the clash point of generation gaps (Zemke, et. al., 2000). Jeffries and Hunte (2003) also affirm that all generational cohorts have different value systems and respond uniquely to situations, and the knowledge of generational characteristics provides an understanding of the diversified workforce as well as the personal motivators of employees and presents a challenge for organizational leadership when navigating a diversified workforce.

This present study sought to evaluate the extent to which tenure and position category moderate generational cohorts' readiness for organizational change in the healthcare industry at healthcare systems across the Southeastern United States, to clarify the relationship between generational cohorts, and to extend the literature on the generational theory and organizational change. The present study employed a large heterogeneous sample size to ensure a geographically diverse sample across the

Southeastern United States of America. This research focused on generational healthcare employees who experienced an organizational change in the past or are currently involved with a change initiative. Additionally, this study focused on four of the five generations (Baby Boomers, Generation X, Generation Y, and Generation Z), as these groups constitute most of the modern workforce.

# Significance of the study

Kitchen and Daly (2002) suggest that the study of organizational change is essential and defines organizational behavior. Additionally, the increasing generational diversity of the modern workforce requires an understanding of the influence of generational characteristics on organizational change. However, there is limited research on the change readiness of generational cohorts in the healthcare industry. Therefore, this study significantly contributes to the narrative and literature on organizational change readiness and change management strategies. This dissertation may help close the gap in current literature relevant to the generational theory and organizational change readiness theory and individual response to organizational change. This dissertation presents insight not previously provided that highlights the relationship between generational cohorts and change readiness in the healthcare industry.

Further, this research provides insight lending to the understanding of generational response to organizational change. The data gathered from this study may equip healthcare administrators with knowledge of how different generations respond to organizational change initiatives; the healthcare sector may be better equipped to implement change initiatives successfully. This study may extend the literature on organizational change and reveal the role that generational cohorts play in employee

change readiness when moderated by tenure and position descriptions. The information gathered in this research may provide healthcare leaders with insight on how to adapt change initiatives to meet the characteristics of different generations.

### **Discussion of findings**

The purpose of this study was to determine the extent to which tenure level and position category moderate generational cohorts''readiness for organizational change in the healthcare industry at healthcare systems across the Southeastern United States. This current research sought to examine the relationship among generational cohort characteristics and readiness for organizational change and to determine if this relationship, if any, is moderated by employee tenure level or position category. The correlational analysis was guided by the Omnibus research question: *What is the relationship between generational cohorts (GENCO) and employee readiness for organizational change (ROC) in a healthcare environment when moderated by tenure (TE) and position category (POC)?* 

A correlational analysis was used to examine the relationship of the variables of the two research questions that emerged from the Omnibus research question. The two research questions are: (1) To what extent do the generational cohorts predict a relational effect on employee readiness for organizational change? and (2) To what extent do the interaction of tenure, position category, and generational cohorts predict a relational effect on employee readiness for organizational change? It was hypothesized that generational cohort characteristics would not have a statistically significant effect on employee readiness for organizational change ( $H_01$ ), and the influence of TE and POC on generational cohorts would not predict a statistically significant relationship in employee

readiness for organizational change ( $H_02$ ). The results of these correlational analyses are discussed in the following section.

#### Hypothesis one (H<sub>0</sub>) and Hypothesis two (H<sub>0</sub>1)

The first and second hypotheses sought to examine how generational cohort characteristics explain the relationship between readiness for organizational change and when correlated with the variables of tenure level and positions category. Using Pearson's correlation coefficient, the quantitative correlational analysis showed a positive correlational relationship between generational cohort characteristics and change readiness. The analysis also showed a positive correlational relationship when the position category was introduced as a variable, but the significantly negative correlational relationship between the tenure level and generational characteristics. Although a positive relationship exists between generational cohort characteristics and change readiness, the strength of the correlation, apart from the negative correlation, did not demonstrate a statistically significant relationship. As such, the null hypothesis, which stated that generational cohorts would not explain the relationship between the dependent variable (readiness for organizational change), and the independent variable (generational cohorts), was supported. Moreover, tenure levels showed an inverse relationship with generational cohorts and change readiness.

# Negative relationships: TE, GENCO, and ROC

A negative correlational relationship reveals that two variables are moving in different or opposite directions; here, an increase in one variable (TE) is associated with a decrease in the second variable (GENCO) and vice versa (Rogers & Nicewander, 1988). A significantly negative correlation was observed when the tenure level (TE) variable

was correlated with the generational cohorts (GENCO) to examine the influence of generational tenure level on change readiness (ROC). However, this correlation was not proportional. The relationship suggests that an increase in the length of employment does not have a proportional relationship with the generational cohort age group when organizational change readiness is considered. Since an inverse relationship exists with GENCO and TE, the result suggested that the four generations (BABO, GENX, GENY, iGEN) responded to change initiatives independently irrespective of their length of employment. Here, the significantly negative correlation indicates that as one moves up the generational cohort scale, the tenure levels of employees decreases and becomes less significant.

## Hypothesis three (H<sub>0</sub>2)

Pearson's correlation coefficient test was used to evaluate the variables relating to hypothesis 3. Analysis of the data supporting hypothesis 3, which examined the extent to which tenure and position category interact with generational cohorts to have a relational effect on readiness for organizational change dimensions, demonstrated the existence of linear relationships. As mentioned above, there was a significantly negative correlation between the tenure of work in health care organizations and generations which signified that the two variables are inversely related. Additionally, a positive relationship existed between generational cohort characteristics and the three dimensions of organizational change readiness: intention, cognition, and emotion. The correlational analysis showed a positive relationship between generational cohorts, intentional readiness, cognitive readiness, and emotional readiness. However, the demonstrated relationships were not significant, and as such, the null hypotheses were supported.

The results revealed that the positive relationship between the generational cohort and the position category was not significant. The finding indicates that the position level of generational employees did not have a significant influence on their readiness for organizational change. Additionally, the results showed that the emotions demonstrated towards change initiatives, the attitude towards the change, and the willingness to engage in the change was not significantly different from one generation to the other. The findings from this current study showed that all generational cohorts surveyed did not rely on their emotions when making decisions about the initiated organizational change. This was evident because there was no significant relationship between generational cohorts and emotional readiness for change. Further, the current research also revealed that the attitude or thought process of the generational cohorts was not significantly different. There was no significant relationship between generational cohorts and cognitive change readiness. Finally, the results indicated that the four generational cohorts surveyed demonstrated a similar propensity to exert energy to engage in the change process, as evidenced by the lack of significant relationship between intentional change readiness and generational cohorts.

#### Summary

After the review of the data, the overall perception presented by the quantitative correlational research was that generational cohort characteristics have a positive linear relationship with readiness for organizational change. However, the perceived linear relationship is very weak, and therefore, statistically not significant. Additional analysis performed to determine the impact of generational cohort characteristics was performed using regression analysis. The analysis revealed that there was no association between the

independent variable (GENCO), the mediated variables (TE and POC), and readiness for organizational change (ROC); this is because a moderate correlation between the variables was not established. As such, a linear relationship could not be determined. Further, an Analysis of Variance (ANOVA) test was performed to determine the variance and difference between generational cohorts and three dimensions of change readiness. While the cognitive and intentional change readiness dimensions were found to have no significant variance or difference from the average mean, a significant variance was observed for the emotional dimension of change readiness.

# Literature support

As discussed in Chapter 2, the literature has demonstrated that organizational change initiatives elicit cognitions, emotions, and intentions that inform employees' response and behavior toward change and also have a bearing on how they feel, think, or act in relation to their role and impact within the organization and, consequently, their readiness for change (Armenakis et al., 1993; Bouckenooghe et al., 2009; Oreg et al., 2003; Oreg, et al., 2018). However, the results generated from this study did not show a statistically significant relationship between generational employees' readiness for organizational change. While generational characteristics and perceptions differ and impact the response to change according to the literature (Pihulyk, 2003; Price & Chahal, 2005), correlational evidence of generational cohort influence on change readiness could not be determined, as the relationship observed was not significant. However, because the present study indicates that there is a significant variance between generational cohorts and the emotional dimension of change readiness, further studies might be warranted to

investigate this phenomenon since the high cost and failure rate of change initiatives persist.

## **Implications of the study**

Readiness for organizational change is a multi-faceted construct that allows organizational members and the organization to share a common commitment to implement change (Weiner, 2009). The successful implementation of change within organizations carries individual and organizational level implications. This current study will allow organizations to clearly identify and define the need for change, formulate the intended change, and assess the readiness of individual organizational members. Although not significant, the current study indicates that generational cohort characteristics positively correlate with the dimensions of organizational change readiness. Understanding of the generational characteristics will foster cooperative behaviors while allowing organizations to attain the knowledge of change efficacy. "When organizational members share a common, favorable assessment of task demands, resource availability, and situational factors, they share a sense of confidence that collectively they can implement a complex organizational change. In other words, change efficacy is high" (Wiener, 2009, p. 4). Here, individual organizational members and the organization could benefit as a result of knowledge acquisition and creation as a result of implementing change initiatives.

There is an overall perception that generational differences influence the ability to embrace organizational change. Generational resistance to change initiatives has been said to increase along the generational scale with the iGeneration cohort less resistant to organizational change and the Baby Boomers more resistant to change (Aldisert, 2002;

Becton, et al., 2014; Bursch & Kelly, 2014; Holtshouse, 2010; Singh, 2013; Valcour, 2013). Although differences may exist between these generations, the findings from this current research did not identify a significant relationship between generational cohorts and their resistance or acceptance to initiated organizational change. As such, the current study indicates that there is no significant relationship or difference in generational readiness for change. This study implies that the Baby Boomer generation, Generation X, Millennials, and iGeneration will all respond equally to the change initiative implemented by organizations due to the lack of significant relationship between generational cohorts and change readiness. As healthcare leaders strive to manage organizational changes such as the shift in consumer behavior patterns, regulatory adjustments, IT developments, and the increasing consolidation of the healthcare industry, generational employee responses to these changes is one less barrier to overcome. Finally, an implication as a result of this current study is that the Baby Boomer generation, Generation X, Millennials, and iGeneration all respond equally with the same thought process, emotion, and attitude when organizational change is initiated or implemented.

### Limitations of the study

The scope of this quantitative correlational research was limited to the number of healthcare employee participants across the Southeastern states of the United States of America. This is a limitation because the self-reported survey instrument used to collect the data cannot guarantee the authenticity and veracity of the study participants. Additionally, this quantitative correlational research included limitations imposed by the survey instrument because surveys can limit the accuracy of the data. The data used for the study originated from the Likert-scaled Organizational Change Questionnaire (OCQ-

C, P, R) survey instrument. The five-point scale used descriptive words such as strongly disagree, disagree, neither agree nor disagree, agree, and strongly agree. Three of the survey questions were reversed scale. It is possible that participants did not consider the reverse scaled questions when giving responses. This limitation could have resulted in a variance in the dataset which cannot be accounted for examining the relationships among variables.

Another potential limitation involved the disparity of the frequency of gender in the dataset. The distribution of females in the sample significantly exceeded those of males and was not representative of employed healthcare employees across the Southeastern states of the United States. Along with gender, a potential limitation involved the geographic distribution of the population. This sample was collected from a population that was concentrated in eight Southeastern states of the United States, a population that is not representative of the United States of America. Selecting participants according to the gender and geographic distribution indicated in the U.S. Census Bureau population data could have averted this limitation. Finally, this quantitative correlational study was delimited to adult healthcare employees working in the Southeastern United States.

# Recommendations

Organizational change continues to receive increased attention in recent years; organizations continue to experience difficulties when managing an increasing rate of change programs (Kotter, 2002; Saka, 2003). Additionally, with four generations (Baby Boomers, Generation X, Generation Y, and Generation Z) with different characteristics, personality traits, ethnicity, culture, and life experiences now working side-by-side
(Lancaster & Stillman, 2002; Jeffries & Hunte, 2003, White, 2006), future research on organizational change should include all four generations. This current study should be repeated with a larger sample size and a better distribution of gender from all regions of the United States and globally. Additionally, due to the large composition of the female gender in this study, future studies should evaluate the relationship between females within the generational cohorts and readiness for organizational change. It may be beneficial to include both qualitative and quantitative methods when this research is repeated.

More studies are needed to determine the impact and influence of generational cohort characteristics and readiness for organizational change. Future studies to understand if the cultural diversity of generational cohorts predict organizational change readiness, and the extent to which the cultural diversity influence employee readiness for organizational change is needful to expand the organizational change and generational theory literature. Such a study will increase the understanding of the individual, employee level change readiness when the ethnicity of generational cohorts is considered. Further, it might be necessary to understand whether or not personality traits moderate or predict organizational change readiness and is, therefore, recommended. Additionally, it might be needful to apply a qualitative methodology to observe the behavior of different generations under various organizational change conditions, allowing for a more in-depth understanding of the lived experiences of generational cohorts.

#### Conclusions

This quantitative, cross-sectional research was significant because it provided insight into the relationship among four generational cohorts and three dimensions of

98

readiness for organizational change. The findings of the present study were that positive relationships exist between generational cohorts and the three dimensions of change readiness; however, the existing relationships were not statistically significant. Study findings also revealed that there is a difference between the generational cohort characteristics and the emotional change readiness dimension. The findings from this study may be useful to organizational leaders, enabling them to develop an awareness of generational characteristics and differences when developing effective organizational change strategies to ensure successful change implementation within the healthcare environment.

#### REFERENCES

- Abdel-Ghany, M. M. M. (2014). Readiness for change, change beliefs and resistance to change of extension personnel in the New Valley governorate about mobile extension. *Annals of Agricultural Science*, 59(2), 297-303.
- Adcroft, A., Willis, R., & Hurst, J. (2008). A new model for managing change: The holistic view. *The Journal of Business Strategy*, 29(1), 40–45. doi:10.1108/02756660810845697
- Aguinis, H. (1995). Statistical power problems with moderated multiple regression in management research. *Journal of Management*, 21, 1141–1158. doi:10.1016/0149-2063(95)90026-8
- Aguinis, H., & Gottfredson, R. K. (2010). Best-practice recommendations for estimating interaction effects using moderated multiple regression. *Journal of Organizational Behavior*, 31, 776–786. doi:10.1002/job.686
- Aldisert, L. (2002). Valuing people: How human capital can be your strongest asset.Chicago, IL: Dearborn Trade Publishing.
- Appelbaum P.S. (2007) Assessment of patient's competence to consent to treatment. *New England Journal of Medicine*, 357: 1834-1840.
- Armenakis, A.A. Harris, S.G. (2002). Crafting a change message to create transformational readiness, *Journal of Organizational Change Management*, Vol. 15 Issue: 2, pp.169-183, https://doi.org/10.1108/09534810210423080
- Armenakis, A. A., Harris, S. G., & Mossholder, K. W. (1993). Creating readiness for organizational change. *Human Relations*, 46, 681–703. doi:10.1177/001872679304600601

- Aguinis, H., & Pierce, C.A. (2006). Computation of effect size for moderating effects of categorical variables in multiple regression. *Applied Psychological Measurement*, 30, 440–442. doi:10.1177/0146621606286314
- Akan, B., Er Ülker, F., & Ünsar, A. S. (2016). The Effect of Organizational Communication Towards Resistance to Change: A Case Study in Banking Sector. *Economic Review: Journal of Economics & Business / Ekonomska Revija: Casopis Za Ekonomiju i Biznis*, 14(1), 53–67.
- Avey, J.B., Wernsing, T.S., & Luthans, F. (2008). Can positive employees help positive organizational Change? Impact of psychological capital and emotions on relevant attitudes and behaviors. *Journal of Applied Behavioral Science*, 44(1), 48–70. doi:10.1177/0021886307311470
- Axley, S. R., & McMahon, T. R. (2006). Complexity: A frontier for management education. *Journal of Management Education*, 30, 295–315. doi:10.1177/1052562905277316
- Azen, R., & Budescu, D. (2009). Applications of multiple regression in psychological research.
  In R. Millsap, & A. Maydeu-Olivares (Eds.), *The Sage handbook of quantitative methods in psychology*. (pp. 285-311). London: Sage. doi:10.4135/9780857020994.n13
- Bareil, C. (2013). Two paradigms about resistance to change. *Organization Development Journal*, 31(3), 59–71.
- Barnett, W., & Carroll, G. (1995). Modeling internal organizational change. *Annual Review of Sociology*, 21, 217–236. doi:10.1146/annurev.so.21.080195.001245
- Bateh, J., Castaneda, M. E., & Farah, J. E. (2013). Employee resistance to organizational change. International Journal of Management & Information Systems (Online), 17, 113–116.

Retrieved from http://www.cluteinstitute.com/journals/internationaljournal-ofmanagement-information-systems-ijmis/

- Becker, H. S. (1960). Notes on the concept of commitment. *American Journal of Sociology*, 66, 32–40
- Becker, M.C., Lazaric, N., Nelson, RR., & Winter, S.G. (2005). Applying organizational routines in understanding organizational change. *Industrial & Corporate Change*, 14, 775–791. doi:10.1093/icc/dth071
- Becton, J.B., Walker, H. J. & Jones-Farmer, L.A. (2014). Generational differences in workplace behavior. Journal of Applied Social Psychology, 44(3): 175-189. doi: 10.1111/jasp.12208
- Beer, M., & Walton, A. E. (1987). Organization change and development. Annual Review of Psychology, 38, 339-367
- Benzer, Charns, Hamdan, & Afable (2017). The role of organizational structure in readiness for change: A conceptual integration. *Health Services Management Research*, Vol. 30(1), 34-46.
- Bless, H. (2001). Mood and the use of general knowledge structures. In L. Martin & G.Clore (Eds.), *Theories of mood and cognition: A user's handbook* (pp. 9-26).Mahwah, NJ: Lawrence Erlbaum Associates, Inc.
- Blythe, J., Baumann, A., Zeytinoglu, I. U., Denton, M., Akhtar-Danesh, N., Davies, S., Kolotylo, C. (2008). Nursing Generations in the Contemporary Workplace. *Public Personnel Management*, 37 (2), 137-159
- Bommer, W. H., Rich, G. A., & Rubin, R. S. (2005). Changing attitudes about change: longitudinal effects of transformational leader behavior on employee cynicism about

organizational change. *Journal of Organizational Behavior*, 26, 733–753. doi:10.1002/job.342

- Bouckenooghe, D. (2010). Positioning change recipients' attitudes toward change in the organizational change literature. *The Journal of Applied Behavioral Science*, 46(4), 500-531. https://doi.org/10.1177/0021886310367944
- Bouckenooghe, D., Devos, G., & Broeck, H. v. d. (2009). Organizational change questionnaireclimate of change, processes, and readiness: Development of a new instrument. The *Journal of Psychology*, 143, 559–599. doi:10.1080/00223980903218216
- Bourne, B. (2015). Phenomenological Study of Generational Response to Organizational Change. *Journal of Managerial Issues: JMI; Pittsburg*, 27(1–4), 141-159,8.
- Braz, P., Frey, E., Rohr da Cruz, M., Camargo, M., & Olea, P. (2011). Consumption of generation "Z" stratified from human Maslow's needs. *International Journal of Management and Administrative Sciences*, 1(2), 1.
- Burke, W. W. (2013). *Organization change: Theory and practice (4th ed.)*. Thousand Oaks, CA: Sage.
- Burnes, B. (2004). Managing change. London, England: Prentice Hall.
- Burnes, B. (2005). Complexity theories and organizational change. *International Journal of Management*, 7(2), 73-90.

Burnes, B. (2015). Understanding resistance to change–building on Coch and French. Journal of Change Management, 15(2), 92-116. https://doi.org/10.1080/14697017.2014.969755

Burnes, B. & Jackson, P. (2011). Success and failure in organizational change: An exploration of the role of values. *Journal of Change Management* Vol. 11, No. 2, 133–162.

- Bursch, D., & Kelly, K. (2014, November). Managing the multigenerational workplace. UNC Kenan-Flagler Business School, Executive Development. Retrieved from https://www.kenan-flagler.unc.edu
- Cabrera, J. (2017). Millennials, Generation Z and the US Army recruiting efforts: Innovative approaches to reach the newest generations yet. (Master's Thesis). Savannah College of Art and Design. Retrieved from http://ecollections.scad.edu/iii/cpro/DigitalItemViewPage.external?sp=1003864
- Caliskan, S. & Isik, I. (2016). Are you ready for the global change? Multicultural personality and readiness for organizational change, *Journal of Organizational Change Management*, Vol. 29 Issue: 3, pp.404-423, https://doi.org/10.1108/JOCM-07-2015-0119
- Cândido, C., & Santos, S. (2015). Strategy implementation: What is the failure rate? *Journal of Management & Organization*, 21(2), 237-262. doi:10.1017/jmo.2014.77
- Choi, M. & Ruona, W.A. (2011). Individual readiness for organizational change and its implications for human resource and organization development. *Human Resource Management*, 10(1), 46–73. doi:10.1177/1534484310384957
- Chung, S.-H., Su, Y.-F., & Su, S.-W. (2012). The Impact of Cognitive Flexibility on Resistance to Organizational Change. *Social Behavior & Personality: An International Journal*, 40(5), 735–745.
- Clore, G., Wyer, R., Dienes, B., Gasper, K., Gohm, C., & Isbell, L. (2001). Affective feelings as feedback: Some cognitive consequences. In L. Martin & G. Clore (Eds.), *Theories of mood and cognition: A user's handbook* (pp. 27-62). Mahwah, NJ: Lawrence Erlbaum Associates, Inc.

- Cohen, J. (1988). *Statistical Power Analysis for the Behavioral Sciences (2nd ed.)*. Hillsdale, NJ: Lawrence Erlbaum Associates, Publishers.
- Cohen, J. (1992). Quantitative methods in psychology: A power primer. *Psychological Bulletin*, 112(1), 155-159.
- Cohen, A. (1991). Career stage as a moderator of the relationships between organizational commitment and its outcomes: A meta-analysis. *Journal of Occupational Psychology*, 64, 253–268.
- Cohen, A. (1992). Antecedents of organizational commitment across occupational groups: A meta-analysis. *Journal of Organizational Behavior*, 13, 539–558.
- Cone, J. D., & Foster, S. L. (2004). Dissertations and theses from start to finish. Washington, DC: *American Psychological Association*.
- Cooper, D. R., & Schindler, P. S. (2003). *Business research methods* (8th ed.). Boston: McGraw-Hill.
- Corsten, M. (1999). The time of generations. Time & Society, 8(2-3), 249-272.
- Creswell, J. W. (2005). Educational research: Planning, conducting, and evaluating quantitative and qualitative research (2nd ed.). Columbus, OH: Pearson.
- Crow, G. Wiles, R. Heath, S., & Charles, V. (2006). Research ethics and data quality: The implications of informed consent. *International Journal of Social Research Methodology*, 9(2), 83-95.
- Decker, P., Durand, R., Mayfield, C. O., McCormack, C., Skinner, D., & Perdue, G. (2012).
   Predicting implementation failure in organization change. *Journal of Organizational Culture, Communications & Conflict*, 16(2), 39–59. Retrieved from http://www.alliedacademies.org/Public/Default.aspx

- Dimock, M. (2019, January 17). Defining generations: Where Millennials end and Generation Z begins. Retrieved February 1, 2019, from <u>http://www.pewresearch.org/fact-</u> <u>tank/2019/01/17/where-millennials-end-and-generation-z-begins/</u>
- Farag, A., Tullai-McGuinness, S., & Anthony, M. (2009). Nurses' perception of their Manager's leadership style and unit climate: are there generational differences? *Journal of Nursing Management*, 17(1), 26-34. doi:10.1111/j.1365 2834.2008.00964.x
- Faul, F., Erdfelder, E., Buchner, A., & Lang, A.-G. (2013). Statistical power analyses using G\*Power 3.1: Tests for correlation and regression analyses. *Behavior Research Methods*, 41, 1149-1160.

Field, A. (2009). Discovering statistics using SPSS (3rd ed.). Thousand Oaks, CA: Sage.

- Ford, J. D., & Ford, L. W. (2010). Stop blaming resistance to change and start using it. *Organizational Dynamics*, 39(1), 24–36. doi:10.1016/j.orgdyn.2009.10.002
- Foster, K. (2013). Generation and discourse in working life stories. *The British Journal of Sociology*, 64(2), 195-215. doi:10.1111/1468.4446.12014
- Gall, M. D., Gall, J. P., & Borg, W. R. (2003). *Educational research: An introduction* (7th ed.). Boston: Allyn & Bacon.
- Gallo, A. (2015). A refresher on Regression analysis. *Harvard Business Review*. Retrieved from <a href="https://hbr.org/2015/11/a-refresher-on-regression-analysis">https://hbr.org/2015/11/a-refresher-on-regression-analysis</a>
- Grama, B., & Todericiu, R. (2016). Change, Resistance to Change and Organizational Cynicism. *Studies in Business & Economics*, *11*(3), 47–54. <u>https://doi.org/10.1515/sbe-2016-0034</u>
- Glass, A. (2007). Understanding generational differences for competitive success. *Industrial and Commercial Training*, 39(2), 98-103.

- Goldberg, L. R. (1993). The structure of phenotypic personality traits. *American Psychologist*, 48(1), 26–34. doi:10.1037/0003-066X.48.1.26
- Hammer, M. and Champy, J. (1993) *Reengineering the corporation: A manifesto for business revolution*. London, England: Nicholas Brearly.
- Higgs, M., & Rowland, D. (2005). All changes great and small: Exploring approaches to change and its leadership. *Journal of Change Management*, 5(2), 121–151. doi:10.1080/14697010500082902
- Hill, J. W., & Powell, P. (2009). *The national healthcare crisis: Is eHealth a key solution*.Business Horizons: 52(3), 265-277.
- Hill, K. S. (2004). Defy the decades with multigenerational teams. *Nursing Management*, 35(1), 32-35. doi:10.1097/00006247-200401000-00012
- Himmelstein, D. U., Woolhandler, S., Almberg, M., & Fauke, C. (2018). The U.S. health care crisis continues: A data snapshot. *International Journal of health Services*, 48(1), 28-41.
- Hobbs, F., & Stoops, N. (2000). U.S. Census Bureau, Census 2000 special reports, seriesCENSR-4. Demographic trends in the 20th Century. Washington, DC: U.S.Government Printing Office.
- Hoffman, M. (1986). Affect, cognition and motivation. In R. Sorrentino & E. T. Higgins (Eds.), Handbook of motivation and cognition (pp. 244-280). New York: The Guilford Press.
- Holt, D. T., Amenakis, A.A., Feild, H.S., & Stanley, Harris, S.G. (2007). Readiness for organizational change: The systematic development of a scale. *The Journal of Applied Behavioral Science*, 43, 232–255. doi:10.1177/0021886306295295

Holtshouse, D. (2010). Knowledge work 2020: Thinking ahead. On the Horizon, 18, 193

203. doi:10.1108/10748121011072645

- Horovitz, B. (2012, May 4). After Gen X, Millennials, what should next generation be. USA *Today*. Retrieved from http://usatoday30.usa
- Hughes, M. (2011). Do 70 percent of all organizational change initiatives really fail? *Journal of Change Management*, 11, 451–464. doi:10.1080/14697017.2011.630506
- Jacobs, G., Witteloostuijn, A.V., & Christe-Zeyse, J. (2013 A theoretical framework of organizational change. *Journal of Organizational Change Management*. Vol. 26 No. 5, pp. 772-792.
- Jefferies, F. L., & Hunte, T. L. (2004). Generations and motivation: A connection worth making. *Journal of Behavioral and Applied Management*, 6(1), 37-70.
- Johnson, M., & Johnson, L. (2010). *Generations, Inc.: From boomers to linksters Managing the friction between generations at work*. New York, NY: AMACOM.
- Karp, T., & Helgø, T. I. (2008). From Change Management to Change Leadership: Embracing Chaotic Change in Public Service Organizations. *Journal of Change Management*, 8(1), 85–96. doi:10.1080/14697010801937648
- Kidwell, R. E. Jr. (2003). Helping older workers cope with continuous quality improvement. *Journal of Management Development*, 22(10), 890-905
- Kitchen, P. J., & Daly, F. (2002). Internal communication during change management. *Corporate Communications*, 7(1), 46–53. doi:10.1108/13563280210416035
- Kotter, J. P. & Cohen, D.S. (2002). *The Heart of Change*. Boston, MA: Harvard Business School Press.
- Kotter, J. P. (2007). Leading change. Harvard Business Review, 85(1), 96-103

- Kotter, J.P. (1996). Leading change. *Competitive Intelligence Review*, 8(2), 96–97. doi:10.1002/cir.3880080221
- Kotter, J. P., & Schlesinger, L. A. (2008). Choosing strategies for change. *Harvard Business Review*, 57(2), 106-14. https://doi.org/10.1007/978-1-137-16511-4\_10
- Kuiper, S. (2008). Introduction to Multiple Regression: How Much Is Your Car Worth? *Journal* of *Statistics Education*, *16*(3), https://doi.org/10.1080/10691898.2008.11889579

Laerd Statistics. (2013). One-way ANOVA in SPSS. Retrieved from https://statistics.laerd.com

- Lamm, E., & Gordon, J.R. (2010). Empowerment, predisposition to resist change, and support for organizational change. *Journal of Leadership & Organizational Studies*, 17, 426–437. doi.10.1177/1548051809355595
- Lancaster, L. C., & Stillman, D. (2002). *When generations collide: Who they are, why they clash, how to survive the generational puzzle at work.* New York: Harper Business.
- Lanier, K. (2017). 5 things HR professionals need to know about Generation Z: Thought leaders share their views on the HR profession and its direction for the future, *Strategic HR review*, 16(6), 288-290. Retrieved from https://doi org.contentproxy.phoenix.edu/10.1108/SHR-08-2017-0051
- Lehmann, E.L & Romano, J.P. (2005). *Testing statistical hypotheses*. New York, NY: Springer Science+Busines Media.
- Lesser, E., & Rivera, R. (2006). Closing the generational divide. IBM Global Business Services. Retrieved January 16, 2019, from http://astd2007.astd.org/PDFs/Handouts%20for%20Web/new%20handouts/M204 .pdf

- Lewin, K. (1947). Frontiers in group dynamics: Concept, method and reality in social science; social equilibria and social change. *Human Relations*, 1(5), 3–41. doi:10.1177/00187267470010010
- Lewis, L. K. (2006). Employee perspective on implementation communication as predictors of perceptions of success. Western Journal of Communications, 70(1), 23-46.
- Ludviga, I., & Sennikova, I. (2016). Organisational change: generational differences in reaction and commitment. Vilnius, LITHUANIA: VGTU Press. https://doi.org/10.3846/bm.2016.10
- Mannheim, K. (1952). *Essays on the sociology of knowledge*. New York: Oxford University Press.
- Marshall, J. (2004). Managing different generations at work. Financial Executive, 18.
- Marston, C. (2009). Myths about Millennials: Understand the myths to retain Millennials. Retrieved from http://humanresources.about.com/od/management tips/a/millennial\_myth.htm
- Martin, C. A., & Tulgan, B. (2006). *Managing the generation mix: From urgency to opportunity*. Amherst, MA: HMD Press.
- Miller, D. (2002). Successful change leaders: What makes them? What do they do that is different? *Journal of Change Management*, 2, 359–368. doi:10.1080/714042515
- Moore, D. S. (2001). *Statistics: Concepts and controversies (5th ed.)*. New York: W.H. Freeman and Company.
- Mowday, R. T., Porter, L. W., & Steers, R. M. (1982). *Employee organizational linkages: The psychology of commitment, absenteeism, and turnover*. New York: Academic Press
- Murphy, S. (2007). Leading a multigenerational workforce. Washington, DC: AARP.

- Nelson, B. (2007). Tips and techniques to bridge the generation gap. *Health Care Registration*, 3-5.
- Nimon, K.F., & Oswald, F.L. (2013). Understanding the results of multiple linear regression:
   Beyond standardized regression coefficients. *Organizational Research Methods*, 16(4), 650–674. doi:10.1177/1094428113493929.
- Nohe, C., Michaelis, B., Menges, J. I., Zhang, Z., & Sonntag, K. (2013). Charisma and organizational change: A multilevel study of perceived charisma, commitment to change, and team performance. *The Leadership Quarterly*, 24(2), 378–389.
  doi:10.1016/j.leaqua.2013.02.001
- Nunnally, J.C. (1977). Psychometric theory. New York, NY: McGraw-Hill.
- Oreg, S. (2003). Resistance to change: Developing an individual differences measure. *Journal of Applied Psychology*, 88(4), 680–693. doi:10.1037/0021-9010.88.4.680
- Oreg, S., Bartunek, J. M., Lee, G., Do, & Do, B. (2018). An Affect-Based Model of Recipients' Responses to Organizational Change Events. *Academy of Management Review*, 43(1), 65– 86. https://doi.org/10.5465/amr.2014.0335
- Oreg, S., & Sverdlik, N. (2011). Ambivalence toward imposed change: The conflict between dispositional resistance to change and the orientation toward the change agent. *Journal of Applied Psychology*, 96(2), 337–349. doi:10.1037/a0021100
- Oreg, S., Vakola, M., & Armenakis, A. (2011). Change recipients' reactions to organizational change: A 60-year review of quantitative studies. *Journal of Applied Behavioral Science*, 47(4), 461–524. doi:10.1177/0021886310396550

Pallant, J. (2010). *SPSS survival manual* (Kindle version). McGraw Hill International Parry, E. (2017). *Generational diversity at work: New research perspectives*. Abingdon,

United Kingdom: Routledge.

- Pedhazur, E.J. (1997). *Multiple regression in behavioral research: Explanation and prediction,* (*3rd ed.*). Fort Worth, TX: Harcourt Brace.
- Pellettiere, V. (2006). Organization self-assessment to determine the readiness and risk for a planned change. *Organization Development Journal*, 24(4), 38–43. Retrieved from http://www.highbeam.com
- Piderit, S. K. (2000). Rethinking resistance and recognizing ambivalence: A multidimensional view of attitudes toward an organizational change. *Academy of Management Review*, 25(4), 783-794.
- Pihulyk, A. (2003). Embrace change: Understand and overcome resistance to change. *Canadian Manager*, 28(4), 27-28.
- Poole, P.P. (1998). Words and deeds of organizational change. *Journal of Managerial Issues*, 10(1), 45-59
- Price, A. D. F., & Chahal, K. (2006). A strategic framework for change management. *Construction Management and Economics*, 24, 237-251
- Rawlins, C., Indvik, J., & Johnson, P. R. (2008). Understanding the new generation: What the Millennial cohort absolutely, positively must have at work. *Journal of Organizational Culture, Communications and Conflict, 12*(2), 1–8.
- Sanzgiri, J., & Gottlieb, J. Z. (1992). Philosophic and pragmatic influences on the practice of organizational development 1950-2000. *Organizational Dynamics*, 21(2), 57-69.

- Schweiger, S., Stouten, H., & Bleijenbergh, I. L. (2018). A System Dynamics Model of Resistance to Organizational Change: The Role of Participatory Strategies. *Systems Research and Behavioral Science*, 35(6), 658–674. https://doi.org/10.1002/sres.2509
- Sessa, V. I., Kabacoff, R. I., Deal, J., & Brown, H. (2007). Generational differences in leader values and leadership behaviors. *The Psychologist Manager Journal*, 10(1), 47-74.
- Seung-Bum, Y., & Guy, M.E. (2006). GenXers versus Boomers work motivators and management implementations. *Public Performance and Management Review*, 29(3), 267-284.
- Sharma, N., Hernschmidt, J. Claes, V., Batchnick, S., De Geest, S., & Simon, M. (2018).
  Organizational readiness for implementing change in acute care hospitals: An analysis of a cross-sectional, multicentre study. *Journal of Advance Nursing*,74:2798–2808.
- Shin, J., Taylor, M., & Seo, M. (2012). Resources for change: The relationships of organizational inducements and psychological resilience to employees' attitudes and behaviors toward organizational change. *Academy of Management Journal*, 55(3), 727–748. doi:10.5465/amj.2010.0325
- Sinclair, M. (2013). The 'z generation': Digital mothers and their infants. *Evidence Based Midwifery*, 11(1), 3.
- Singh, P. (2013) Millennials and the workplace: Challenges for architecting the organizations for tomorrow *Human Resource Management International Digest*, Vol. 21 Issue: 7, https://doi.org/10.1108/HRMID.04421gaa.003
- Skår, S., Sniehotta, F. F., Araújo-Soares, V., & Molloy, G. J. (2008). Prediction of behaviour vs. prediction of behaviour change: The role of motivational

moderators in the theory of planned behaviour. *Applied Psychology: An International Review*, 57, 609-627. doi:10.1111/j.1464-0597.2008.00346.x

- Smith, M. E. (2002), Implementing organizational change: Correlates of success and failure. *Performance Improvement Quarterly*, 15(1), 67–83. doi:10.1111/j.19378327.2002.tb00241.x
- Snedecor, G.W & Cochran, W.G. (1989). *Testing methods*. Ames, IA: Iowa State University Press.

Stanton, J. M. (2001) Galton, Pearson, and the Peas: A Brief History of Linear Regression for Statistics Instructors, *Journal of Statistics Education*, 9:3, DOI: 10.1080/10691898.2001.1191053

- Staw, B., & Barsade, S. (1993). Affect and managerial performance: A test of the sadder but-wiser vs. happier-and-smarter hypothesis. *Administrative Science Quarterly*, 38, 304-331.
- Stets, J. E., & Burke, P. J. (2000). Identity theory and social identity theory. Social Psychology Quarterly, 63(3), 224-237.
- Storms, A. (2004). How to ease tension when changes occur. *San Diego Business Journal*, 25(41), 33
- Strauss, W., & Howe, N. (1991b). The cycle of generations. *American Demographics*, 13(4), 24 31.
- Strauss, W., & Howe, N. (1992). Generations: This history of America's future, 1584 to 2069. New York: William Morrow and Company, Inc
- Strebel, P. (2009). Why do employees resist change? *IEEE Engineering Management Review*, 37(3), 60–66. doi:10.1109/emr.2009.5235497

- Swanson, R.A & Holton III, E.F. (2005). *Research in organizations: Foundations and methods of inquiry (Editors)*. San Francisco, CA: Berrett-Koehler.
- Szabla, D. B. (2007). A multidimensional view of resistance to organizational change: Exploring cognitive, emotional, and intentional responses to planned change across perceived change leadership strategies. *Human Resource Development Quarterly*, 18(4), 525–558. doi:10.1002/hrdq.1218
- Taber, K. S. (2016). The Use of Cronbach's Alpha when developing and reporting research instruments in science education. *Cross-Mark*, 1-24, doi: 10.1007/s11165-016-9602-2
- Taverkol, M. & Dennick, R. (2011). Making sense of Cronbach's alpha. International Journal of Education, 2:53-55, doi:10.5116/ijme.4dfb.8dfd
- Tulgan, B. (2004, Winter). Trends point to a dynamic generational shift in the future workforce. *Employment Relations Today*, 23-31
- Tulgan, B. (2009). Not everyone gets a trophy: How to manage Generation Y. San Francisco,CA: Jossey-Bass.
- Twenge, J. (2006). Generation Me. New York, NY: Free Press.
- United States Census Bureau. (2017). Retrieved from http://www.census.gov/acs/www/
- Vakola, M., & Nikolaou, I. (2005). Attitudes towards organizational change: What is the role of employees' stress and commitment? *Employee Relations*, 27(2), 160–174, doi:10.1108/01425450510572685
- Valcour, M. (2013, December 19). Coaching is key to engaging generation Y. *HR Review*. doi:10.13140/RG.2.1.3568
- Van de Ven, A. H., & Poole, M. S. (1995). Explaining development and change in organizations. Academy of Management Review, 20(3), 510–540.

- Van de Ven, A. H., & Sun, K. (2011). Breakdowns in Implementing Models of Organization Change. Academy of Management Perspectives, 25(3), 58–74. https://doi.org/10.5465/amp.25.3.zol58
- Van Egeren, L.F. (2009). A cybernetic model of global personality traits. *Personality and Social Psychology Review*, 13(2), 92–108. doi:10.1177/1088868309334860
- Vogt, P.W. (2007). *Quantitative research methods for professionals*. Boston, MA: Allyn and Bacon.
- Warrick, D.D. (2009). Developing organization change champions. *OD Practitioner*, 41(1), 14 19. Retrieved from http://www.odnetwork.org/publications/practitioner/index.php
- Weick, K. E., & Quinn, R. E. (1999). Organizational change and development. Annual Review of Psychology, 50(1), 361–386. doi:10.1146/annurev.psych.50.1.361
- Weiner, B. J. (2009). A theory of organizational readiness for change. *Implementation Science*, 4(1). https://doi.org/10.1186/1748-5908-4-67
- Weston, M. (2006). Integrating generational perspectives in nursing. Online Journal of Issues in Nursing, 11(2), 12.
- White, W. (2006, November). Four generations learning to work better together. *Public Management*, 35
- Wright, T.A. and Bonett, D. G. (2002). The moderating effects of employee tenure on the relation between Organizational Commitment and Job Performance: A Meta-Analysis. *Journal of Applied Psychology* Vol. 87, No. 6, 1183–1190. DOI: 10.1037//0021-9010.87.6.1183

- Yang, Y., Choi, J. N., & Lee, K. (2018). Theory of Planned Behavior and Different Forms of Organizational Change Behavior. Social Behavior & Personality: An International Journal, 46(10), 1657–1671.
- Zedeck, S. (1971). Problems with the use of "moderator" variables. *Psychological Bulletin*, 76, 295–310. doi:10.1037/h0031543
- Zemke, R., Raines, C., & Filipczak, B. (2000). *Generations at work: Managing the clash of veterans, Boomers, Xers, Nexters in your workplace*. New York, NY: AMACOM
- Zwick, T. (2002) Employee resistance against innovations. *International Journal of Manpower*,
   Vol. 23 Issue: 6, pp.542-552, https://doi.org/10.1108/01437720210446397

## APPENDICES

#### **Appendix A: Informed consent**

#### **INFORMED CONSENT DETAILS**

**Research Topic:** The relationship among generational cohorts, tenure, job categories, and employee readiness for organizational change in a healthcare environment: A Quantitative Study

#### **INTRODUCTION**

The purpose of this informed consent form is to provide you with information that may affect your decision to participate or not participate in this research. This consent form also records the consent of those who agree to be involved in the study.

#### RESEARCH

Jerry S.K. Adatsi is a student researcher at George Fox University's College of Business in the Doctor of Business Administration Program. He has invited you to participate in a research study investigating the relationship among healthcare employees of different generations and their readiness for organizational change.

#### **STUDY PURPOSE**

The purpose of this research is to determine if employees' generational cohort characteristics contribute to their readiness for organizational change. The results of this study will allow for a greater understanding of factors that contribute to organizational change readiness in the healthcare sector.

This is an online study intended to identify the difference among generations (Baby boomers, Generation X, Generation Y, and Generation Z) and the readiness for organizational change within a healthcare system organizational change effort. This research will explore how generational characteristics, employee tenure, and position categories of an employee impacts the readiness for change.

#### ELIGIBILITY

You are eligible to participate in this research if you:

1. Are currently employed at a healthcare organization.

2. Are at least 18 years of age.

You are not eligible to participate in this research if you:

- 1. Are not currently employed at a healthcare organization
- 2. Are not at least 18 years of age.

#### **DESCRIPTION OF RESEARCH ACTIVITY**

If you agree to participate in the study, you will be asked to:

Participate in an online survey. You will be asked to take the survey at the time and location of your choosing. Each participant will approximately 7 minutes taking the online survey. Approximately 120-500 participants will be participating in this research study.

#### RISK

There are no known risks or foreseeable risk from participating in this study. In any study, there is some possibility that you may be subject to risks that have not yet been identified. To decrease the potential for risks, you can stop participation in the survey at any time.

#### BENEFIT

You will receive no direct benefits from participating in this research study. Your responses may help us learn more about the employee's readiness for organizational change and generational cohorts within healthcare organizations.

Although individual employees of your organization will not directly benefit from participating in this study, a summary of the study findings, without any individual responses, will be made available to your organization's leadership at the conclusion of the study.

#### **CONFIDENTIALITY AND ANONYMITY**

All information collected in this study is strictly confidential and anonymous unless disclosure is required by law. What you say, and how you answer the questions in this survey cannot be connected to you, and no one will know if you participated or not in this study. The results of this research may be used in reports, presentations, and publications. The researcher, Jerry S.K. Adatsi, will not identify individual participants at any point.

To maintain confidentiality and anonymity of your records, Jerry S.K. Adatsi will not collect personal identifying information with the exception of gender, education level, job position, and length of employment. Data will only be reported as a summary of all data collected. The survey data will be stored in a password protected location. The only person who will have access to the information you provide is the researcher. Your information will be secured in a password protected computer file on a password protected computer.

Your survey answers will be collected on Qualtrics.com where data will be stored in a password protected electronic format. Qualtrics does not collect identifying information such as your name, email address, phone number or IP address.

The data collected will be kept for a maximum of 5 years. At that point, the electronic data will be destroyed.

#### WITHDRAWAL PRIVILEGE

Participation is voluntary. You have the right to decline to participate in this research study. If you agree to participate now, you are free to withdraw later. You can also stop participating at any point while taking the survey. There are no penalties to you for not participating.

If you decide to withdraw from the study, your employment status will not be affected. To stop participation in the survey, you may exit the online survey at any time. If you decide to stop participation, the information collected from you will not be used in the study.

#### **COSTS AND PAYMENTS**

There is no financial cost to you as a participant in this study. There is no payment for your participation.

#### COMPENSATION FOR ILLNESS AND INJURY

Your consent to participate in this study does not waive any of your legal rights. No funds have been set aside to compensate you in the event of injury.

#### VOLUNTARY CONSENT

You are invited to ask questions if you have concerns about the study or about your participation in the study. Please contact Jerry S.K. Adatsi with any questions by email at georgefox.edu.

You are invited to ask any questions about your rights as a participant in this research or voice concerns if you feel you have been placed at risk. Please contact the Chair of the Internal Review Board at George Fox University College of Business by email at @georgefox.edu.

This form explains the nature, demands, benefits, and risk of the research study. By clicking "I consent, begin the study" you confirm that you are 18 years or older, understand the content of this form, and agree to participate in this study.

I Consent, begin the study \_\_\_\_\_

I do not consent, I do not wish to participate

#### Appendix B: Readiness for organizational change survey

#### **Demographic questions**

The following are demographic questions about yourself. This information is not shared with any third party and cannot be connected to you. Please answer the following questions.

#### 1. Are you 18 years of age and older?

- Yes
- O<sub>No</sub>

# 2. Are you an employee of a hospital or medical clinic in Southeastern, United States (Georgia, Tennessee, Alabama, Mississippi, Louisiana, North Carolina, and South Carolina)?

- O Yes
- <sub>No</sub>

### 3. In what year were you born?

- <sup>O</sup> 1946 to 1964
- <sup>O</sup> 1965 to 1980
- <sup>O</sup> 1981 to 1996
- 1997 to Later

### 4. Which position category best describe your job function?

<sup>O</sup> Medical Support Staff (Doctors, Nurses, and Technicians staff)

<sup>O</sup> Administrative Staff (Executives, Managers, Supervisors, Finance, HR, and organization operations staff)

Specialty and Ancillary Services Staff (Laboratory, cardiology, and other lay staff)

<sup>O</sup> Data Management and Other Staff (IT, HIM, and Other organizational staff not mentioned above)

### 5. How long have you worked at your current company?

- $\circ$  0 2 years
- 3 5 years
- 6 8 years
- $\circ$  9+ years

### 6. What is your education Level

| 0  | Associates Level |  |  |  |
|--|------------------|--|--|--|
| 0  | Bachelor's Level |  |  |  |
| 0  | Master's Level   |  |  |  |
| 0  | Doctoral Level   |  |  |  |
| <b>7. What is your gender?</b><br>Female |                  |  |  |  |

<sup>O</sup> Male

#### Intentional readiness for change (IRC) questions

This part contains questions about specific change within your department or organization. We are interested in finding out about people's attitudes to change. In answering the following questions, please have the specific change project in mind. Especially try to remember those things that particularly affected you and your immediate colleagues. Select an answer on a scale of 1 through 5 (1-Strongly Disagree, 2-Disagree, 3-Neutral, 4-Agree, and 5-Strongly Agree)

#### 8. I want to devote myself to the process of change

| • Strongly<br>Disagree   | O<br>Disagree | • Neither agree or disagree<br>Neutral | O<br>Agree | • Strongly<br>Agree |  |
|--|---------------|--|------------|---------------------|--|
| 9. I am willing to make a significant contribution to the change |               |  |            |                     |  |
| • Strongly   | 0             | • Neither agree or                     | 0          | • Strongly          |  |
| Disagree   | Disagree      | disagree                               | Agree      | Agree               |  |
| 10. I am willing to put energy into the process of change        |               |  |            |                     |  |
| Strongly   | 0             | • Neither agree or                     | 0          | O Strongly          |  |
| Disagree   | Disagree      | disagree                               | Agree      | Agree               |  |

#### Cognitive readiness for change (CRC) questions

This part contains questions about specific change within your department or organization. We are interested in finding out about people's attitudes to change. In answering the following questions, please have the specific change project in mind. Especially, try to remember those things that particularly affected you and your immediate colleagues. Select an answer on a scale of 1 through 5 (5-Strongly Disagree, 4-Disagree, 3-Neutral, 1-Agree, and 1-Strongly Agree)

# 11. I think that most changes will have a negative effect on the client/patient we serve

| O Strongly  | 0        | • Neither agree or | 0     | • Strongly |  |
|---|----------|--------------------|-------|------------|--|
| Disagree  | Disagree | disagree           | Agree | Agree      |  |
| 12. Plans for future improvement will not come to much  |          |                    |       |            |  |
| O Strongly  | 0        | • Neither agree or | 0     | Strongly   |  |
| Disagree  | Disagree | disagree           | Agree | Agree      |  |
| 13. Most change projects that are supposed to solve problems around here will not<br>do much good |          |                    |       |            |  |

| Strongly | 0        | • Neither agree or | 0     | • Strongly |
|----------|----------|--------------------|-------|------------|
| Disagree | Disagree | disagree           | Agree | Agree      |

#### **Emotional readiness for change (ERC) questions**

This part contains questions about specific change within your department or organization. We are interested in finding out about people's attitudes to change. In answering the following questions, please have the specific change project in mind. Especially try to remember those things that particularly affected you and your immediate colleagues. Select an answer on a scale of 1 through 5 (1-Strongly Disagree, 2-Disagree, 3-Neutral, 4-Agree, and 5-Strongly Agree)

#### 14. I have a good feeling about the change project

| O Strongly                       | 0             | Neither agree or   | 0     | Strongly   |  |  |
|----------------------------------|---------------|--------------------|-------|------------|--|--|
| Disagree                         | Disagree      | disagree           | Agree | Agree      |  |  |
| 15. I experience                 | the change as | a positive process |       |            |  |  |
| <sup>O</sup> Strongly            | 0             | • Neither agree or | 0     | • Strongly |  |  |
| Disagree                         | Disagree      | disagree           | Agree | Agree      |  |  |
| 16. I find the change refreshing |               |                    |       |            |  |  |
| O Strongly                       | 0             | Neither agree or   | 0     | Strongly   |  |  |
| Disagree                         | Disagree      | disagree           | Agree | Agree      |  |  |

#### Appendix C: Permission for the use of copyright content



Jerry Adatsi < @georgefox.edu>

# Request to use OCQ-CPR for dissertation research

**Dave Bouckenooghe** @brocku.ca> < To: "Jerry S.K. Adatsi" < @georgefox.edu>

Thu, Jan 31, 2019 at 10:59 AM

You have permission to use the OCQ-CPR.

Best regards,

Dave B.



Jerry Adatsi < @georgefox.edu>

### Request to use OCQ-CPR for dissertation research

Jerry S.K. Adatsi < @georgefox.edu> @brocku.ca To:

Wed, Jan 30, 2019 at 6:37 PM

Good evening Dr. bouckenooghe,

My name is Jerry S.K. Adatsi, and I am a doctoral student at George Fox University. I am currently beginning my dissertation focusing on organizational readiness for change and generation cohorts. I came across your measurement tool for readiness for organizational change while performing my literature review. I am writing to ask for your permission to use the OCQ-CPR measurement tool for my research. The details of the study are provided in a previous email below. I would much appreciate your correspondence on this matter and the opportunity to use your measurement tool.

Thank you,

Jerry S.K. Adatsi

With My Warmest Regards,

I Remain. Jerry S.K. Adatsi, MBA | Doctor of Business Administration (DBA) Student (ABD - All But Dissertation) George Fox University | College of Business | @georgefox.edu